

Symposium on Social Interactions in Complex Intelligent Systems (SICIS)

In conjunction with the 2018 Convention of the Society for the Study of Artificial Intelligence and Simulation of Behaviour (AISB 2018)

6th April 2018









Adaptation and the Social Salience Hypothesis of Oxytocin: Early Experiments in a Simulated Agent Environment

Imran Khan, Matthew Lewis and Lola Cañamero¹

ABSTRACT

Allostasis is a mechanism that permits adaptation of an organism as a response to changing (physical or social) environmental conditions. Allostasis is driven by a number of factors, including regulation through hormonal mechanisms. Oxytocin (OT) is a hormone that has been found to play a role in regulating social behaviours and adaptation. However, the concrete effects that OT promotes remain unclear and controversial. One of these effects is on the attention paid to social cues (social salience). Two opposing hypotheses have been proposed. One hypothesis is that adaptation is achieved by increasing attention to social cues (increasing social salience), the other that adaptation is achieved by decreasing attention to social cues (decreasing social salience). In this paper, we present agent simulation experiments that test these two contrasting hypotheses under different environmental conditions related to food availability: a comfortable environment, a challenging environment, and a very challenging environment. Our results show that, for the particular conditions modelled, increased social salience through the release of simulated oxytocin presents significant advantages in the challenging conditions.

I INTRODUCTION

Endowing artificial (robotic or simulated) agents with the ability to "survive" in constantly-changing environmental conditions poses interesting challenges. In areas such as autonomous robotics and human-robot interaction, robots must be able to both anticipate and adapt to new conditions, in order to adapt to realistic, dynamic physical and social environments. Affect is one of the mechanisms that permits adaptation (Panksepp, 2004). Through this adaptation, expression and emergence of affect-based behaviour will become more context-relevant, able to evolve over time, adapting to changes whenever necessary (Schulkin, 2011).

In previous work undertaken in our research group (e.g., Avila-Garcia and Cañamero, 2002; Attwood et. al., 2013; Cañamero and Lewis, 2016; Lewis and Cañamero, 2016), regulation of the internal environment (milieu) has largely been through homeostatic processes. Homeostasis describes the process that an agent undergoes in order to maintain its internal milieu at a stable level, for example through error-correcting behaviours. In these models, the internal milieu is corrected through negative feedback loops once "physiological" deficits occur. Anticipation of changing environmental circumstances is difficult when regulation is approached in this way.

A more anticipatory adaptive process is allostasis (Schulkin, 2011). Allostasis describes the regulatory process that an agent undergoes to adjust its internal milieu when presented with, and even in anticipation of, a change in environmental conditions. Given influences from or changes in the external environment, allostasis allows for the range of values of the internal milieu to change dynamically - a fundamental difference with homeostasis, where this range doesn't change. Whilst homeostatic regulation is effective in static and some types of dynamic environments, it is limited in its approach, particularly in conditions where the range of the internal milieu needs to be adapted to accommodate changes in the environment, for example to accommodate changes in temperature under different seasons. Allostasis can therefore be seen as an adaptation of the mechanism that permits adaptation to the environment, and proposes longer-term viability (Ashby, 1952) of agents through mechanisms of adaptation and anticipation.

Allostatic process can be triggered as a response to changes in either the physical environment (Sterling, 1988) or the social environment (Schulkin, 2011), in which case it is called social allostasis. Social allostasis describes the mechanism of allostasis taking into account the social environment by using social behaviours to regulate the internal conditions. Schulkin (2011) states that social regulation of internal conditions is a fundamental behavioural adaptation, and that the evolution of allostasis was largely driven by social adaptation. Hormones are one of the mechanisms that are involved in social allostasis., Within the social allostasis literature, different hormones have been mentioned (Bethelem et. al., 2014). One such hormone that has consistently been found to play a role in this social regulation is oxytocin (De Dreu et. al., 2016). Oxytocin (OT) is a neuromodulatory hormone that has been gaining increasing attention in recent years due to its presumed facilitation of prosocial behaviours in mammals (Neumann, 2009, Chang et. al., 2012, Krueger et. al., 2013). Recently, Ebitz and Platt (2014) argue that prosocial behaviours may not always be the most optimal approach when considering an agent's overall viability and adaptation to a social environment. In their paper, these authors cite several examples of population-level simulation studies where success of the population was driven by antisocial behaviours such as deception and defection. They therefore hypothesised that oxytocin should reduce social vigilance, rather than enhance it, in order to contribute to the viability of social agents.

OT has also been found to be context-dependent (Bartz et. al., 2011; Bos et. al. 2012), and has also been shown to drive more seemingly antisocial behaviours, such as defensiveness towards out-group members (De Dreu et. al., 2010, 2011) and envy (Shamay-Tsoorey et. al., 2009), despite also being heavily correlated with prosocial behaviours such as bonding (Williams et. al, 1994), reward-sharing in macaques (Chang et. al., 2012), and parental care (Gordon et. al., 2010). Due to the observed facilitation of social

¹ Embodied Emotion, Cognition and (Inter-)Action Lab, School of Computer Science, University of Hertfordshire, U.K. email: i.khan9@herts.ac.uk, m.lewis4@herts.ac.uk, l.canamero@herts.ac.uk

behaviours, the mechanisms of oxytocin provide a good initial framework on which to build models of social regulation and social allostasis. However, the precise effects that oxytocin has on social behaviours within biological agents is still not unanimous, with opinions divided as to whether prosocial behavious equates to a more viable life. What this provides is an opportunity to test some of the hypothesised effects of OT, and to use these findings as a framework to build a social allostatic model upon in future research.

In robotics, very little work has been done so far modelling OT as a mechanism for social adaptation. For example, Anumandla et al. (2011) modelled OT release within a robot in order to assess the impact on trust in human-robot interactions. Although less directly related to our work, since it doesn't involve the use of OT, it is also worth mentioning work on social adaptation of robots such as Zahadat et al. (2014), who studied the social adaptation of robots through a combination of hardware configurations and evolutionary algorithms.

As an initial study into this long-tailed research of social allostasis, the experiments in this paper investigate the effects of the prosocial hypothesis of the mechanisms of oxytocin, modelled using the NetLogo simulation framework (NetLogo, 2001). We have modelled a small society of agents under conditions implementing two opposing effects on social salience: either increasing attention to social cues, i.e., increasing social salience (Gordon et. al., 2010), or decreasing attention to social cues, i.e., decreasing social salience (Ebitz and Platt, 2014). The aim of our study is to investigate if mechanisms of oxytocin that modulate social salience would contribute, or be detrimental, to the overall viability and survival of agents under different conditions in our simulated environment.

The remainder of the paper is structured as follows. Section 2 discusses the agent model of social salience and its action selection architecture. Section 3 describes the metrics used to assess the results of the experiments. Section 4 describes the experimental methodology. Section 5 presents the results of our experiments and a discussion of our findings. Finally, section 6 summarizes our conclusions and proposes future work.

2 AGENT MODEL

2.1 INTERNAL VARIABLES

In all experiments, agents have two homoestatically controlled physiological variables – *Nutrition* and *Socialness*. Both variables have identical lower limits (set to 0), upper limits (set to 1) and ideal values (set to 1) and are controlled through a homeostatic process using an architecture inspired by the work done by Avila-Garcia and Cañamero (2002). The value of both *Nutrition* and *Socialness* must thus be within a range between 0 and 1; they are satiated by the eating of resources and grooming with other agents, respectively. What is important to note is that, although both internal variables (*Nutrition* and *Socialness*) drive error-correcting behaviours, only a maximum error of the *Nutrition* variable can cause the agent to die. In other words, an agent cannot die of *loneliness*, but can die of *hunger*. The drive to correct the socialness variable deficit still drives the *loneliness* motivation and the grooming behaviour, despite the possibility of death. Drives are

Physiological Variables Update Rate		
Nutrition	-0.0003	
Socialness	-0.0003	
Oxytocin (when enabled)	-0.0005	

Table 1. Rate of change, per time step, of each internal variable for each agent. Rates have been determined through a series of preexperimental runs in order to create enough difference between conditions to investigate.

calculated concurrently for each agent at each time step. All agents are subject to the same rate of decay for each of their internal variables. The decay rate for each internal variable can be found in Table 1.

2.2 AGENT MOVEMENT

The default movement pattern of an agent is a random wander through the world, until that wandering behaviour is interrupted by another behaviour or perception of resource. Agents perceive the world through a cone of vision of length 20 and 80 degrees (40/+40) and, for the sake of simplicity, are able to determine the distances, resource type and resource colour without requiring any additional cognition or computation. However, this may not be the case in future experiments.

2.3 AGENT BEHAVIOURS

Aside from the appetitive behaviour of wandering, agents have the following two consummatory behaviours:

Eat: During the execution of this behaviour, the agent will stop at a food resource and begin consuming it. For each "bite" (which is updated every time step) the agent recovers its nutrition at a rate of 0.01. At the same time, the food resource reduces in size at a rate of -0.01 - an inverse of the physiological effect on the agent. Therefore, the nutritional value that is taken from the resource is equal to the value that is consumed from it.

Groom: Grooming is achieved through tactile feedback of another agent. When *groom* is executed, agents who pass each other and are within the defined "touching" distance will satisfy that behaviour through touching. Grooming lasts as long as is necessary for the error to be corrected to a point where the motivational state changes (when *hunger* becomes the motivation). The *robot-radius* for *grooming* was set at 2 after initial tests of different radius sizes led to negligible differences, and is the value that is included in this model.

		Effect on Physiology (Per Time Step)		
Behaviour b _i	Stimuli	Nutrition	Socialness	Oxytocin
Eat	Food	+ 0.01	0.0	0.0
Groom	Agent	0.0	+ 0.05	+1

Table 2. Effects on internal variables for both of the possible consummatory behaviours the agent can execute. Again, rates have been determined through a series of pre-experimental runs to create enough variance for investigation.

Table 2 describes the effects that the successful execution of each of the consummatory behaviours has on the internal variables of the agent. As we will see in the next section, these effects are used during the calculation of behaviour intensity in the penultimate step of the action-selection architecture described below.

The choice of having asymmetrical effects of both behaviours was not random. Initially, both physiological effects (*Nutrition* and *Socialness*) were set at the same values (+0.01) in order to create symmetrical effects for both behaviours. However, modelling this into the agents resulted in no differences across all conditions. Conversely, bringing the physiological effect of *groom* up to 0.1 (ten times the effect of *eat*) did bring changes, but these were too small to analyse in detail. After experimenting with these values, the physiological effect for *grooming* behaviour was set at +0.05 to the *socialness* variable. As *eat* is satisfied by consuming static resources, and *groom* is satisfied by a consummatory behaviour on dynamic resources, the relative difficulty of satisfying that drive should be offset by an asymmetrical rate of satisfaction.

2.4 ACTION-SELECTION ARCHITECTURE

The action-selection architecture is a motivated architecture following the model in (Cañamero 1997). In addition to the abovementioned physiological variables and behaviours, the architecture uses motivations that combine the intensity of the internal needs set by the physiological variables, and the observed external stimuli, in order to select the behaviour that the agent needs to execute. The particular way to calculate motivations and winning behaviours was inspired by the winner-takes-all motivated architecture described in (Avila-Garcia and Cañamero, 2002). Table 3 summarizes how these different elements of the architecture are connected, showing a breakdown of each internal variable, with the motivation it drives, the resulting behaviour that is executed by that motivational drive, the stimulus that is affected by that consummatory behaviour and the resulting physiological effect on the internal variables.

The action selection algorithm, running continuously in a loop, can be described as follows. For both behaviours, at each time step:

 Calculate the *drive/error of each internal variable*. Errors are calculated as the difference between the current values of the variables from their ideal value. At each time step, for each variable, v:

$$e_{\nu} = p_{\nu} - j_{\nu} \tag{1}$$

where e_v is the error value, p_v is the ideal value for that internal variable, and j_v is the current value of that internal variable.

- Calculate the *cues, i.e., the value of each external stimulus*. The value of *food cues* is calculated as:

$$C_{food} = \Sigma nutritionValue$$
(2)

where Σ *nutritionValue* is the total nutritional amount of food resources seen in an agent's cone of vision.

Internal Variable	Motivation	Behaviour	Stimuli	Physiological Effect
	m_i	b_i	S	f
Nutrition	Hungry	Eat	Food	+ nutrition
Socialness	Lonely	Groom	Agent	+ socialness

Table 3. Relationships between internal variables and motivations, through to the effects on physiology of both consummatory behaviours.



Figure 1. Overview of the action-selection architecture as described. The order of computation is from left to right, and both drives are calculated concurrently.

The value of robot cues is calculated as:

$$C_{robot} = \Sigma \, robots \tag{3}$$

Where Σ robots is the *count* of the number of other agents seen in an agent's vision.

Calculate the *salience of each stimulus*. The value of *food salience* is the same as *food cues*, therefore:

$$S_{food} = C_{food} \tag{4}$$

The value of *social salience* is calculated as:

$$S_{robot} = C_{robot} \times \alpha$$
 (5)

where α is the effect of oxytocin defined in the next section. Calculate the individual *intensity of both motivations*, *hungry* and *lonely*. Motivation is calculated as follows:

$$m_i = e_v + (e_v + S_i)$$
(6)

where e_v is the error of internal variable v and S_i is the stimuli linked to that motivation. The motivation with the highest intensity is selected as the winning motivation:

$$m_{winner} = max[m_{hungry}, m_{lonely}]$$
 (7)

- Calculate the *intensity of both behaviours* by multiplying the winning motivation by the physiological effect of each of the behaviours:

$$= m_{winner} \times f_{i_{winner}} \tag{8}$$

where m_{winner} is the winning motivation and $f_{i_{winner}}$ is the physiological effect that m_{winner} has on behaviour b_i .

- The behaviour with the highest intensity is the winning behaviour, and the one that is executed.

$$b_{winner} = max[b_{eat}, b_{aroom}] \tag{9}$$

Figure 1 above gives a visual representation of the architecture.

2.5 EFFECTS OF OXYTOCIN

Within the context of this work, oxytocin is described as a hormone that affects, through a modulation function, the salience of social cues. *Salience* is defined here as the likelihood of a stimulus being given attention. The definitions of each effect are as follows:

- None: Oxytocin does not have any effect on social salience
- Increased Social Salience: Social salience is increased proportional to the oxytocin within the agent:

$$S_{robot} = C_{robot} \times (1 + Oxytocin) \quad (10)$$

- *Decreased Social Salience:* Social salience is increased proportional to the oxytocin within the agent:

$$S_{robot} = C_{robot} \times (1 - Oxytocin)$$
 (11)

3 METRICS

3.1 DEFINITIONS

Taking note of what Avila-Garcia and Cañamero (2002) stated, caution must be taken in assessing performance on a single indicator of viability. For instance, a longer lifespan is not, in itself, necessarily indicative of greater viability. Additional aspects, such as comfort or life quality, may supplement the understanding of an agent's viability. The definitions of each viability indicator observed in this study are below:

- *Life Length:* The time that the robot survived (remained viable by keeping the *nutrition* variable above 0) during each run, as a percentage of the maximum simulation time. Time is defined as the number of steps in each simulation:

$$S_{life} = \frac{t_{life}}{t_{simulation}}$$
(12)

where t_{life} is the time the agent remained alive and $t_{simulation}$ is the maximum length of the simulation (in this case, 25,000 time steps).

- *Overall Comfort:* The average level of satisfaction of internal variables, measured at each time step.

$$q_t = 1 - \left(\frac{\Sigma \, Err}{max_{Err}}\right) \tag{13}$$

where ΣErr is the sum of errors of the robot's two internal variables at time *t*, and max_{Err} as the sum of the maximum

errors of all the internal variables. The average comfort level per run is given by:

$$Comfort = \frac{\sum q_t}{t_{life}}$$
(14)

Motivational State Balance: The time that each robot spent in each motivational state (either *hungry* or *lonely*), measured as a percentage of their life-length. For each motivation *i*:

$$m\delta_i = \frac{t_i}{t_{life}} \tag{15}$$

where t_i is the time (in time steps) spent with the winning motivation *i*.

- *Behaviour Execution Balance:* The time that each robot spent executing each of the behaviours (either *eat* or *groom*), measured as a percentage of their total time spent executing behaviours.

$$b\delta_i = \frac{t_i}{t_{behaviours}} \tag{16}$$

where t_i is the time (in time steps) spent executing winning behaviour *i* and $t_{behaviours}$ is the total time spent executing behaviours.

3.2 STATISTICAL SIGNIFICANCE TESTING

In order to gain confidence in the findings, tests of statistical significance were performed on the experimental results. The test used was *One-Way ANOVA*, declaring significance at a p-value of $p \le 0.05$. Significance testing was performed on the two key viability indicators – length of life and overall comfort – across all experimental conditions.

4 EXPERIMENTS

4.1 EXPERIMENTAL HYPOTHESIS

Extending upon the hypotheses proposed in previous literature, the null hypothesis for this experiment is as follows:

 H_0 : Increased social salience will not significantly increase viability of agents.

4.2 EXPERIMENTAL METHODOLOGY

In order to study the discussed effect of oxytocin, the experiments in this paper model a society of six homogenous agents within a simulated environment, each equipped with an action-selection architecture, responsible for driving a homeostatic process for two internal variables (*Nutrition* and *Socialness*). Additionally, each agent has an internal value of *Oxytocin*, which either increases or decreases the salience of social cues. This was tested over three different environmental conditions over 180 simulation runs.

4.3 WORLD DESCRIPTION

The simulated world is modelled in NetLogo (2001) as a flat, twodimensional environment with a fixed size of 99 * 99, encapsulated between four blue walls which the agents cannot pass through. If an agent perceives the edges of the world, it turns around in a random direction and continues as normal. Agents perceive a wall if it falls within max-vision-length / 4 of the agent. Food resources are represented in the simulation as yellow circles, and agents have a priori understanding of what a food resource is. The size of these resources is dynamic - and is dependent on the "nutritional value" available within each food resource (a resource with nutritional availability of 1 will be size 1, for instance). In each world, the nutritional value of each food resources regenerates at a rate equivalent to 1/10,000th of its maximum nutrition value at each time step. Inspiration for this slower rate of regeneration was taken from natural biology, in that – in the real world – the rate at which "natural" food grows is much slower than the rate at which it is consumed.

Initially, six homogenous agents of the same species exist in each world. The agents are visually represented as "bugs" within the simulation of size 3. The starting position of each of the robots is always a random space in the simulation, but is set so that agents do not initialise too close to the food resource. Each experimental condition (as defined in the next section), was run a total of twenty times to a maximum of 25,000 time steps per simulation run. This equated to 180 total experimental runs.

4.4 EXPERIMENTAL CONDITIONS

Each experimental condition is defined by a combination of two variables:

- The conditions of the world (i.e. "easy", "challenging" or "super challenging".)
- The presence, and directional effect, of oxytocin within the action-selection architecture.

The combination of these variables creates nine different experimental conditions to be tested. Each combination, along with its corresponding experiment number for ease of readability in later sections, can be seen in Table 4.

Oxytocin / World Condition	No Oxytocin	Increase Social Salience	Decrease Social Salience
Easy	E0	E1	E2
Challenging	C0	C1	C2
Super Challenging	SC0	SC1	SC2

Table 4. Experimental conditions, defined by the effects of oxytocin on social salience paired against each of the world conditions.



Figure 2 - Screenshots of the model with "easy" conditions (left) and "challenging" conditions (right). "Super challenging" is as "challenging", with smaller resources. Food resources are represented by yellow circles; agents are represented as bugs.

The world conditions (see Figure 2) describe the availability and distribution of food resources in the environment. More specifically:

- *"Easy"* conditions are defined as a world with an abundance of food resources (18 resources), distributed in small clusters throughout the world.
- *"Challenging"* conditions are defined as a world with scarce (6) food resources in two clusters, placed towards the corners of the world.
- *"Super Challenging"* conditions are exactly the same as *Challenging* conditions, although the food resources have half of the nutritional amount.

In each experiment, viability indicators are logged for each individual robot, but reported using the mean results for each experimental condition across all the simulated runs as discussed in a later section. Finally, should all initial food resources be consumed, an additional cluster of four food resources is regenerated in one corner to allow agents to be able to correct their nutrition deficit, albeit with less food availability.

5 EXPERIMENTAL RESULTS

5.1 RESULTS

Regardless of the effects of oxytocin, all agents survived for the full length of all simulation runs during easy world conditions (E0, E1 and E2), resulting in a 100% survival rate across all runs. Despite the consistency in life length, average comfort levels for the easy conditions vary slightly. Little variance (0.3) is seen between E0 (no oxytocin) and E1 (increased social salience), though agents in E2 (decreased social salience) experience a considerably lower average comfort rate. In this case, the easy world had sufficient resources to allow agents to survive, regardless of oxytocin effects.

In the challenging condition, where oxytocin has no effect on the social salience (C0), the life length was, on average for all agents, 75% of the total simulation time. Physiological comfort during these runs was 0.66 (out of a possible 1.0). When oxytocin increased social salience in these challenging conditions (C1), average life length improved to 83%, whilst average comfort levels of agents also increased to 0.71, denoting both a longer and more comfortable life. Decreasing social salience (C2) had the inverse effect, reducing



Figure 3. Average life length (survival) for all agents across all simulation runs in three environmental conditions (Easy (E), Challenging (C) and Super Challenging (SC)) and with three effects of OT (No effect (conditions labelled 0), increased social salience (conditions labelled 1) and decreased social salience (conditions labelled 2)). Bars represent standard error of +/-1 SD. Asterisk denotes statistical significance.



Figure 4. Average comfort levels for each experimental condition. As before, we show results for all three world conditions (E, C and SC) with three effects of OT (0, 1, and 2). Bars represent standard error of +/-1 SD. Asterisk denotes statistical significance.

average life length to 65% - down 10% vs. the control group of no oxytocin, and a 0.11 reduction in average comfort levels to 0.55. Statistical significance was observed across all results for both the length of life and comfort levels. There was a statistically significant difference between SC2 and the other OT conditions, although no significant effect was found between SC0 and SC1. The same is true for average comfort levels in super challenging conditions.

Figure 5 shows the distribution of motivational states (as defined in the Viability Indicators section) across all three oxytocin effects in both world conditions. For E0 and E1, motivational states are fairly balanced, with agents spending a similar amount of time in each of the motivational states. In conditions with decreased social salience (E2), agents appear to be motivated by groom slightly more than in E0 and E1. The more noticeable differences are noted in the challenging conditions. In C0, there is a variance of +/- 2% pts between motivational states, and C1 has a variance of +/- 5% pts; both slightly favouring the motivation to eat. Once again, in conditions where social salience is decreased (C2), the motivation to groom outweighs the motivation to eat by a factor of \sim 5 (83% to 17%). Again, the potential causes for this are discussed in the next section. A similar split is seen for super challenging conditions, with agents in decreased social salience (SC2) being

Motivational State Distribution



Figure 5. Distribution of time spent in each motivational state for each oxytocin effect for all world conditions. Light grey denotes the % time spent under the groom motivation, dark grey is % time spent under the eat motivation.





Figure 6. As figure 5, with the inclusion of time spent executing each behaviour in each motivational state.

heavily skewed towards grooming motivations (78% vs. 22%). Figure 9 shows the time spent executing the behaviours during each motivational state. Any time not spent executing the behaviour is therefore an agent searching for the relevant resource.

5.2 DISCUSSION

Despite Ebitz and Platt's proposed hypothesis that decreased social salience would result in more viable agents, agents in this environment remained more viable when oxytocin increased social salience. Although the model is currently absent of any additional context outside of the maintenance of homeostatic variables, this conclusion is in line with the prosocial hypothesis discussed previously.

The distribution of motivational states across decreased social salience conditions support this stance. Looking at figures 4 and 5, agents would primarily be motivated to eat and, as social stimuli is not salient enough to follow to food resources, would eventually find the resources without any social assistance. Once food has been eaten and the nutrition error has been corrected, agents would then look to satisfy their (non-lethal) social need. As the social error increases over time, the appearance of food resources, despite having

some salience, would eventually not be salient enough to change the motivational state of the agent (calculated by $m_i = e_v + (e_v + S_i)$).

Increased social salience does indeed aid agents in locating food resources, as following an agent that sees a food resource (and whose motivation it is to eat) ultimately draws agents to resources that they may not otherwise have immediately found. However, the more agents that successfully find a food resource cluster, the faster those resources would get consumed. Agents consume food at a fixed rate, regardless of the number of other agents that are currently also at that cluster. In other words, any propensity to share resources does not currently exist within the model. Therefore, whilst oxytocin that increases social salience correlates with viability of agents in this current model, it may not (and, indeed, should not) be the case when food availability becomes sparser. There has been ample literature (e.g., Ebitz and Platt, 2013; Kosfeld et al., 2015; Bethelem et al., 2014) to suggest that the effects of oxytocin should be context-dependent.

The lack of significance in the results from the super challenging conditions can be attributed to the severity of the challenge. With such low food availability in the environment, agents did not have an opportunity to express behaviours driven by the OT modulation. Future studies will reconsider these world conditions.

6 CONCLUSION AND FUTURE WORK

As an initial step towards creating a socially adaptive model, this study has looked into one probable effect of the neuropeptide oxytocin – that it increases social salience – and created a twodimensional simulated system in which to test a previouslyproposed hypothesis of its effects. The agents simulated within this environment were equipped with an action-selection (AS) architecture, inspired by previous AS work done in our research group, with the inclusion of a hormonal modulation function that affects the salience of social cues proportional to the amount of oxytocin in the system. By both increasing and decreasing social salience across different distributions of food resources, the hypothesis proposed by interactionists – that increased social salience would result in a more viable life for agents – was observed in the context of this study.

Whilst this model is a step towards a socially adaptive system, much work still needs to be completed. The development of a model to study social adaptation requires more than homogenous agents with fixed physiological effects of prosocial behaviour. There are further hypotheses which can be modelled, inspired by the findings and observations of this model, and the literature discussed previously.

One of the observed prosocial effects of increased oxytocin is that of reward sharing (Williams et al. 1994), and therefore an opportunity is presented here to extend beyond this study to include opportunities to share the limited resources. Should agents reach food resources at the same time, the role of oxytocin in this system should also be able to determine potential to resource-share. This rate of sharing may be dose-dependent.

Since all agents in this system are homogenous, the effects and modulation of oxytocin remain consistent for all agents and throughout the length of each simulation. This is rarely true in reallife social groups. The likelihood is that social groups do not consist of evenly-matched individuals, and the perception of other individuals of the group is rarely uniform within the society (Mahajan et al., 2011). Both the release and effects of oxytocin has been found to have different effects in non-human primate groups, based on perceived hierarchy (Knight and Mehta, 2014) or the presence of a predatory threat (Ebitz and Platt, 2014; Muroy et al., 2016). Within a homogenous social species with no social threat, the role of oxytocin on social vigilance cannot be properly surmised, and so any findings remain limited. It is hypothesised that the introduction of a perceived stronger species, or stronger agents of the same species, would have a significant impact on agent viability, and this will be addressed in a future study. Additionally, the rate, or dose, of oxytocin released within the system could be affected by a number of factors, such as the time since an agent has last had social interaction, familiarity with a previous grooming partner, or the perceived hierarchical relationship - whether higher or lower between the agents (Knight and Mehta, 2014). It is hypothesised that changing this rate of release from a fixed value to a more contextdependent rate will create more adaptive behaviours which will impact the viability of agents, although the direction of this impact is unknown. Much work must still be undertaken before confidence in these effects can be achieved. Taking into account additional effects of the hormone proposed in the cited literature, more rigorous experiments are planned for the future. The experiments conducted and results of this study have laid the foundations on which to build a more dynamic, socially adaptive model.

REFERENCES

- Anumandla, S.R., Bray, L.C.J., Thibeault, C.M., Hoang, R.V., Dascalu, S.M., Harris, F.C. and Goodman, P.H. (2011). *Modeling oxytocin induced neurorobotic trust and intent recognition in human-robot interaction*. In Neural Networks (IJCNN), The 2011 International Joint Conference on , 3213-3219
- R. Arletti, A. Benelli, A. Bertolini, (1990). Oxytocin inhibits food and fluid intake in rats. In Physiology & Behavior, Vol. 48, Issue 6, 825-830
- Attwood, S., Cañamero, L. and te Boekhorst, R., 2013. SimianWorld A Study of Social Organisation Using an Artificial Life Model. In ECAL, 633-640
- Ashby, W. R (1952). Design for a Brain, Chapman and Hall, London
- Avila-García, O., Hafner, E., Cañamero, L. 2002. Relating Behavior-Selection Architectures to Environmental Complexity. In Proc. 7th Intl. Conf. on Simulation of Adaptive Behavior. Cambridge, MA: MIT Press
- Barraza, J.A. and Zak, P.J. (2009). Empathy toward strangers triggers oxytocin release and subsequent generosity. Annals of the New York Academy of Sciences, Vol. 1167, Issue 1, 182-189.
- Bartz, J., Simeon, D., Hamilton, H., Kim, S., Crystal, S., Braun, A., et al. (2011a) . Oxytocin can hinder trust and cooperation in borderline personality disorder. Soc. Cogn. Affect. Neurosci. Vol. 6, 556–563
- Bartz, J. A., Zaki, J., Bolger, N., and Ochsner, K. N. (2011b). Social effects of oxytocin in humans: context and person matter. Trends Cogn. Sci. Vol. 15, 301–309.
- Bethlehem, R.A., Baron-Cohen, S., van Honk, J., Auyeung, B. and Bos, P.A., (2014). *The oxytocin paradox*. Frontiers in behavioral neuroscience, Vol. 8.
- Bos P. A., Panksepp J., Bluthe R. M., Honk J. V. (2012). Acute effects of steroid hormones and neuropeptides on human social-emotional behavior: a review of single administration studies. Front. Neuroendocrinol. Vol. 33. 17–35
- Cañamero, L.D. (1997). Modeling Motivations and Emotions as a Basis for Intelligent Behavior. In Lewis Johnson, W., (ed.), Proceedings of the

First International Symposium on Autonomous Agents (Agents'97), pp. 148–155. New York, NY: The ACM Press.

- Cañamero, L. and Lewis, M., (2016). Making new "new AI" friends: designing a social robot for diabetic children from an embodied AI perspective. International Journal of Social Robotics, Volume 8, Issue 4, 523-537.
- Chang, S.W., Barter, J.W., Ebitz, R.B., Watson, K.K. and Platt, M.L., (2012). Inhaled oxytocin amplifies both vicarious reinforcement and self reinforcement in rhesus macaques (Macaca mulatta). Proceedings of the National Academy of Sciences, Vol. 109, Issue 3. 959-964.
- De Dreu C. K., Greer L. L., Handgraaf M. J., Shalvi S., Van Kleef G. A., Baas M., et al. (2010). *The neuropeptide oxytocin regulates parochial altruism in intergroup conflict among humans*. Science, Vol. 328. 1408–1411
- De Dreu, C. K. W., Greer, L. L., Van Kleef, G. A., Shalvi, S., and Handgraaf, M. J. J. (2011). Oxytocin promotes human ethnocentrism. Proc. Natl. Acad. Sci. USA, Vol. 108, 1262–1266
- De Dreu, C.K., Kret, M.E., (2016). Oxytocin conditions intergroup relations through upregulated in-group empathy, cooperation, conformity, and defense. Biological Psychiatry, Vol. 79, Issue 3. 165-173.
- Ebitz, R. B., Watson, K. K., and Platt, M. L. (2013). Oxytocin blunts social vigilance in the rhesus macaque. Proc. Natl. Acad. Sci. USA, Vol. 110. 11630–11635
- Ebitz R.B., Platt M.L. (2014). An evolutionary perspective on the behavioral consequences of exogenous oxytocin application. Front. Behav. Neurosci. Vol. 225, Issue 7.
- Gordon, I., Zagoory-Sharon, O., Leckman, J. F., and Feldman, R. (2010). Oxytocin and the development of parenting in humans. Biol. Psychiatry, Vol. 68. 377–382.
- Hemelrijk, C.K. (2000). Social phenomena emerging by self-organisation in a competitive, virtual world ("DomWorld"). In Learning to behave Workshop II: Internalising knowledge. Ieper, Belgium, 11-19.
- Knight, E.L. and Mehta, P.H., (2014). Hormones and hierarchies. In The psychology of social status, Springer New York. Vancouver. 269-301
- Kosfeld, M., Heinrichs, M., Zak, P. J., Fischbacher, U., and Fehr, E. (2005). Oxytocin increases trust in humans. Nature, Vol. 435. 673–676
- Lewis, M. and Cañamero, L., (2016). *Hedonic quality or reward? A study* of basic pleasure in homeostasis and decision making of a motivated autonomous robot. Adaptive behavior, Volume 24, Issue 5, 267-291.
- Krueger, F., Parasuraman, R., Iyengar, V., Thornburg, M., Weel, J., Lin, M., Lipsky, R. H. (2012). Oxytocin receptor genetic variation promotes human trust behavior. Frontiers in Human Neuroscience, Vol. 6, Article 4.
- Lukas, M., Toth, I., Reber, S.O., Slattery, D.A., Veenema, A.H. and Neumann, I.D., (2011). *The neuropeptide oxytocin facilitates prosocial behavior and prevents social avoidance in rats and mice*. Neuropsychopharmacology, Vol. 36, Issue 11. 2159-2168.
- Mahajan, N., Martinez, M.A., Gutierrez, N.L., Diesendruck, G., Banaji, M.R. and Santos, L.R., (2011). *The evolution of intergroup bias:* perceptions and attitudes in rhesus macaques. Journal of personality and social psychology. Vol. 100, Issue 3. 387.
- Muroy, S.E., Long, K.L., Kaufer, D. and Kirby, E.D. (2016). Moderate stress-induced social bonding and oxytocin signaling are disrupted by predator odor in male rats. Neuropsychopharmacology. Vol. 41, Issue 8, 2160-2170.
- Neumann, I.D. (2009), The advantage of social living: Brain neuropeptides mediate the beneficial consequences of sex and motherhood. In Frontiers in Neuroendocrinology. Vol. 30, Issue 4. 483-496
- Panksepp, J., (2004), Affective neuroscience: The foundations of human and animal emotions. Oxford university press.
- Schulkin, J., (2011). Social allostasis: anticipatory regulation of the internal milieu. Frontiers in evolutionary neuroscience. Vol. 111, Issue 2

- Shamay-Tsoory S. G., Fischer M., Dvash J., Harari H., Perach-Bloom N., Levkovitz Y. (2009). *Intranasal administration of oxytocin increases* envy and schadenfreude (gloating). Biol. Psychiatry, Vol. 66. 864–870
- Sterling, P., (2012) Allostasis: a model of predictive regulation. Physiology & behavior, Volume 106, Issue 1, 5-15.
- Wilensky, U. (1999). NetLogo. http://ccl.northwestern.edu/netlogo/. Center for Connected Learning and Computer-Based Modeling, Northwestern University, Evanston, IL.
- Williams, J. R., Insel, T. R., Harbaugh, C. R., and Carter, C. S. (1994). Oxytocin administered centrally facilitates formation of a partner preference in female prairie voles (microtus ochrogaster). J. Neuroendocrinol. Vol. 6, 247–250
- Zahadat, P., Bodi, M., Salem, Z., Bonnet, F., de Oliveira, M.E., Mondada, F., Griparic, K., Haus, T., Bogdan, S., Millsk, R. and Marianok, P., (2014). *Social adaptation of robots for modulating self-organization in animal societies*. In Self-Adaptive and Self-Organizing Systems Workshops (SASOW), 55-60.

How effective is Ant Colony Optimisation at Robot Path Planning

Aaron Wolfenden¹, Neil Vaughan¹

Abstract. This project involves investigation of the problem robot path planning using ant colony optimisation heuristics to construct the quickest path from the starting point to the end. The project has developed a simulation that successfully simulates as well as demonstrates visually through a graphical user interface, robot path planning using ant colony optimisation. The simulation shows an ability to traverse an unknown environment from a start point to an end and successfully construct a route for others to follow both when the terrain is dynamic and static.

1 INTRODUCTION

To be able to prove that a solution to the problem was found and that a good level of success had been achieved, the graphical user interface (GUI) displays, the pheromone trail and the obstacles that occurred which the agents rerouted around, using a grid based output shows the path the agents are taking throughout the solution. This was comparable to other examples of solving the solution with ant colony optimisation (ACO) to see if improvements have been made and whether some ACO variations are better than others depending on the environment.

Robot path planning (RPP) is used within many applications within the technological world, from helping unmanned vehicles to robotics. It is stated by (Cao, 2016) as 'to find a path from the current point (or the start point) to the target point, which is a shortest or a minimum price path without barrier'.

ACO is a heuristic based off the foraging behaviour of ants, originally created by Marco Dorigo and in 1991, and further expanded on in later years. It was originally the 'Ant System' as a basic heuristic for solving optimisation problems such as the Travelling salesman problem (TSP). However later became developed further and more optimised, becoming ACO.

Within the field of artificial intelligence (AI), ACO is becoming increasingly useful. This is especially reinforced when Google, Uber and other companies are experimenting and testing the idea of self-driving cars and thus many career paths are becoming open in the field.

2 BACKGROUND

A literature search was conducted, the scope of which includes the different techniques that have been employed historically to achieve ACO and different variants and modifications that have been made to the algorithm to optimise it in most part to solve the TSP. Furthermore, investigation was made into the area of RPP and the implementation methods that were used to solve this issue.

As well as this, other areas of ACO were investigated such as the usage of artificial neural networks and genetic algorithm hybrid heuristics.

There are some solutions already applied to this problem, such as the artificial potential field method, neural networks, genetic algorithms (GA) and A* searching method as stated by (Yu, Wei, Wang, Ding, & Wang, 2017). However, they went on to state that each of the current solutions does have problems associated to them. There has also been attempts at hybrid models too consisting of different swarm intelligences integrated together.

There have been many different solutions created to try tackle this problem such as genetic algorithms which is the metaheuristic focussing on evolutionary algorithms that are inspired by natural selection and particle swarm optimization which iteratively searches for improvements to a candidate solution. However, the Swarm Intelligence this paper focussed on is ACO which although has proven to work effectively in solving the problem, still has improvements that could be made and investigated.

Another variation of ACO that implements GA into it is smartPath, created by (Châari, Koubâa, Bennaceur, Trigui, & Al-Shalfan, 2012). Within their work they proposed system they state that 'ACO has a stronger local search capability and faster convergence but the algorithm can easily sink into a local optimum' where as GA 'belongs to random optimize processes, so the local convergence problem does not appear; however, this makes its convergence speed slower'. This therefore clearly shows that the shortcomings of one can be improved by the other.

The way they therefore designed the algorithm was so that the initial path is created from using ACO to create a fast-converged optimal path, and the second phase which is using the GA as a way of post optimization to improve the quality of the solution. This is done by checking all the nodes and then attempting to mutate the nodes in the path if the length of a resulting new path is shorter.

When tested against other heuristics such as their improved ACO (IACO) and classic ACO (CACO) and even GA it was found to outperform each of them in a varying number of environments in terms of both finding the shortest path and the efficiency with time.

Another variation on ACO is AntFarm designed by (Collins & Jefferson, 1990). Their idea was to create a simulation of an evolving population of ant colonies, where the reproduction is based on the amount of food they can carry back to nest, thus promoting better foraging strategies. The colonies are made up of identical ants, however their behaviour is specified by an artificial neural network (ANN).

¹ Dept. of Computer Science, Univ. of Chester, CH65 7AL, UK. Email: {n.vaughan}@chester.ac.uk.

This approach is interesting as it implements a genetic algorithm (GA) within the ACO algorithm. With this they aimed to implement natural evolution using local competition and mating. The other main interest within this research is the use of an ANN to represent each ant's behaviour. Through their use of GA, they can mutate the ANN allowing evolution of each colony of ants. The advantage of a technique such as this is that they can mutate and optimise many features of the ants such as pheromone density, how the optimal path is defined and other factors.

3 HEURISTIC DEVELOPMENTS

The way in which the heuristic functions is that the agents (ants) will go out from the start point (nest) and search for the end objective (food) when the ant reaches the food it will return to the nest and lay a track of pheromone. Until the path to the food is discovered the ants will decide on random directions, without any decision making. However, when a food source is discovered and they pick up a pheromone trail is increases the probability that future ants will then take this route. Overtime this trail will strengthen and most agents will follow this trail.

As well as being able to handle single-objective optimization problems where only one food source is available, (Dehuri, Ghosh, & Cho, 2011) states that ACO also excels at multiobjective optimization problems. These are problems where there may be conflicting objectives which solve the same solution. So, two food sources of equal value, and many ways are presented to solve this issue in their literature review.

Improvements of ACO are also becoming more common as (Yu, Wei, Wang, Ding, & Wang, 2017) states the advantages of ACO are 'its strong robustness' and more importantly that it's 'easy combination with other algorithms'. It was also investigated how much further it would optimise ACO with the addition of evolution within the algorithm brought over from GA. (Roach & Menezes, 2008) state this also, putting forward the evolutionary ant colony optimization (EACO). They further state that through using EACO, it's possible to 'give the individual ants a chance to evolve, and thus, the agents themselves can become more optimized' and they further show in their paper that it is much more powerful in dynamic environments than ACO however ACO is more efficient in static environments.

4 ROBOT PATH PLANNING ACO METHODS

Cao (2016) puts forward an improved version of ACO for RPP. Within it they stated that the initial search time takes too long. This is the result of the randomness at which the ants initially search. Therefore (Cao, 2016) uses a pheromone which decreases with distance with the idea that the ant will have a clear motion direction during the initial search.

Another criticism that (Cao, 2016) made is that the pheromone evaporation rate is unchanged throughout the running of the algorithm, which can lead to local convergence if too small, or slow convergence rate if too large. Therefore, they put forward the notion that a dynamic evaporation rate should be used. This is done with setting the evaporation rate high at the start to enhance the global search ability, then lowering it with the number of cycles so that local convergence can happen on an optimal solution quickly. (Yu, Wei, Wang, Ding, & Wang, 2017) also criticise the original ACO for this and suggest an 'adaptive pheromone volatilization coefficient' where they again suggest that evaporation rate should change with the number of cycles.

The heuristic function in previous work was improved upon by (Cao, 2016). This is done by changing the heuristic function so that the distance between two grid points which is normally used is changed so that instead what is used is the distance between the next grid point the ant will move to and the target point if it is known. This idea is also supported by (Yu, Wei, Wang, Ding, & Wang, 2017) where they also improve the heuristic so that the point that is closer to the goal is chosen instead of blindly picking. The advantage of this is that the initial search speed will be greatly improved and give the agents some direction when searching out a path.

However, (Cao, 2016) uses the ant-cycle algorithm described by (Dorigo, Maniezzo, & Colorni, 1996) after a cycle of movements made, optimal and worst solutions are calculated and some of the best solutions are used and the pheromone quantity is updated as a result.

5 IMPLEMENTATION

The methods aimed to optimise the ACO algorithm as much as possible, so the initial work focussed on an approach to firstly work on the core system methods and then expand to optimising performance, and creating a more complex GUI to control parameters.

The system was created using the python programming language. The reasoning behind this choice was the ease to develop in a short space of time with its extensive library support, readable code and finally due to it being interpreted it is much easier to debug. Libraries used within the creation of the program include Matplotlib which aided in the creation of the graph / grid interface. Another package used is the Tkinter package which provides the GUI with the controls to modify the functionality of the simulator, and was the base GUI for the graph to be imbedded within.

The project was also designed within the Pycharm IDE created by Jetbrains. The reasoning behind this decision was that the IDE provided all the tools and functionality required, especially ease of package management and syntax correction.

The ideas presented in this paper have been successfully implemented as a proof-of-concept prototype as shown in Fig. 1.



Figure 1. Ant Colony with two

The ant colony simulation was then evaluated by comparison to results in literature.

6EVALUATION

To evaluate the developed system the best possible way was to firstly run tests using the algorithm implemented and compare the values and outputs to competitors using ACO or other algorithms to solve RPP. It was also worth comparing how the programming language or changes to the metaheuristics are used as these could possible effect performance. A good starting comparison was to compare the results achieved to work presented by (Dorigo, Maniezzo, & Colorni, Ant System: Optimization by a Colony of Cooperating Agents, 1996) in his paper or results found by (Perumal, et al., 2016) and then furthermore to other more recent variations.

7 CONCLUSIONS & FUTURE WORK

In conclusion ACO shows many potential solutions to solve the problem of RPP. Although ACO does show great ability in solving optimization problems such as TSP and RPP it has shortcomings such as falling into local optimum or not converging quick enough. Many improvements can be made not only to the heuristics itself, but also by creating a hybrid solution using a combination with other heuristics.

A further area of study within the field however would be to consider the problem of multi-objective ACO. As a problem that could occur is if two food sources or end points exist, not only would the shortest path to each must be found, but then these would have to be compared to find which is the optimal solution.

Future work could include extending this to other platforms such as embedded systems. Furthermore looking into applying the algorithm with other languages such as C++ would be ideal as this would provide much greater time efficiency due to its compiled nature and also being a low level language.

Finally another area of interest would be the implementation of evolutionary aspects of GA systems so that it further optimises route planning.

REFERENCES

- Cao, J. (2016). Robot Global Path Planning Based on an Improved Ant Colony Alogrithm. Journal of Computer and Communications, 4, 11-19. doi:http://dx.doi.org/10.4236/jcc.2016.42002
- [2] Châari, I., Koubâa, A., Bennaceur, H., Trigui, S., & Al-Shalfan, K. (2012). smartPath: A hybrid ACO-GA algorithm for robot path planning. Evolutionary Computation (CEC), 2012 IEEE Congress on (pp. 1-8). Brisbane, QLD, Australia: IEEE. doi:10.1109/CEC.2012.6256142
- [3] Collins, R. J., & Jefferson, D. R. (1990). AntFarm: Towards Simulated Evolution. Los Angeles: University Of California. Retrieved from https://www.researchgate.net/profile/David_Jefferson/publication/2 551611_AntFarm_Towards_Simulated_Evolution/links/0fcfd5092 8f91e0fda000000.pdf
- [4] Dehuri, S., Ghosh, S., & Cho, S.-B. (2011). Integration of swarm intelligence and artificial neural network. World Scientific.
- [5] Dorigo, M., & Di Caro, G. (1999). Ant colony optimization: a new meta-heuristic. Proceedings of the 1999 Congress on Evolutionary Computation-CEC99 (pp. 1470-1477). Washington: IEEE. doi:10.1109/CEC.1999.782657
- [6] Dorigo, M., Birattari, M., & Stutzle, T. (2006, November). Ant colony optimization. IEEE Computational Intelligence Magazine, I(4), 28-39. doi:10.1109/MCI.2006.329691
- [7] Dorigo, M., Maniezzo, V., & Colorni, A. (1996). Ant System: Optimization by a Colony of Cooperating Agents. IEEE Transactions on Systems, Man, and Cybernetics, Part B (Cybernetics), 26(1), 29-41. doi:10.1109/3477.484436
- [8] Perumal, N., Rashid, R., Elamvazuthi, I., Tageldeen, M. K., Khan, M. K., & Parasuraman, S. (2016). Mobile robot path planning using Ant Colony Optimization. 2016 2nd IEEE International Symposium on Robotics and Manufacturing Automation (ROMA) (pp. 1-6). Ipoh: IEEE. doi:10.1109/ROMA.2016.7847836
- [9] Roach, C., & Menezes, R. (2008). Handling Dynamic Networks Using Evolution in Ant-Colony Optimization. New Frontiers in Applied Artificial Intelligence (pp. 795-804). Wroclaw: Springer. doi:https://doi.org/10.1007/978-3-540-69052-8_83
- [10] Yu, L., Wei, Z., Wang, H., Ding, Y., & Wang, Z. (2017). Path planning for mobile robot based on fast convergence ant colony algorithm. 2017 IEEE International Conference on Mechatronics and Automation (ICMA) (pp. 1493-1497). Takamatsu: IEEE. doi:10.1109/ICMA.2017.8016037

Morphogenetic Engineering For Evolving Ant Colony Pheromone Communication

Neil Vaughan¹

Abstract. This research investigates methods for evolving swarm communication in a simulated colony of ants using pheromone when foriaging for food. This research implemented neuroevolution and obtained the capability to learn pheromone communication autonomously. Building on previous literature on pheromone communication, this research applies evolution to adjust the topology and weights of an artificial neural network which controls the ant behaviour. Comparison of performance is made between a hard-coded benchmark algorithm, a fixed topology ANN and neuroevolution of the ANN topology and weights. The resulting neuroevolution produced a neural network which was successfully evolved to achieve the task objective, to collect food and return it to the nest.

1 INTRODUCTION

This research has developed a model of ant colony swarm intelligence behaviour. The novel aspect is that behaviour of pheromone navigation was not hard coded, as in most implementations, but has evolved using artificial neural networks (ANNs) and an implementation of neurovevolution. Compared to previous research which failed to evolve standard and fixed topology ANNs for ant behaviour (Collins & Jefferson, 1990a), this research produces successful evolution and applies a more comprehensive neuroevolution methodology including complexification and augmentation of ANN topology and weights, as described by NEAT (Stanley, 2004).

Inspired by biological ants, this research aims to provide insights to advance understanding of how pheromone communication evolved in biological organisms. Application of neuroevolutionary computational modelling provides a useful analogy to how brains may have evolved to produce biological organism behaviours.

There are many long standing open questions regarding the evolution of altruism, related to how any why the evolution of cooperation emerged among closely related individuals [Hamilton 1964]. Worker ants (Formicidae) are a perfect example of altruism, as they collect food for the good of the swarm but they get no individual rewards. This computer simulation method can provide new insights into altruism because each colony is only assessed by its fitness as a whole, not that of individual ants.

Pheromone trails can be seen as social memory or swarm memory used by all agents in the colony. The problem is called central place food foraging which is an optimisation problem. The aim of foriaging is to collect as much food as possible and return it to the nest. An ant's food collection consists of two phases: the search for food and retrieval of food back to the nest.

In this respect the problem relates to the new field of morphogenetic engineering (ME). In this task, the core challenge posed by ME is a reverse engineering one: How can the ants' micro-rules be inferred from the system's macro-objectives? (Doursat et al., 2013). In this case the macro objective is to optimise fitness of the swarm by using swarm communication, but the micro rules for each ant to achieve that were not provided in this system and needed to evolve autonomously with no prefabricated design.

2 RELATED WORK

Literature on pheromone communication is described by various key words: ant evolution simulation pheromone, central-place foraging algorithm (CPFA), pheromone recruitment (Letendre and Moses, 2013).

The core interest of this work is how ant pheromone communication can be evolved in a computational model. There have been some interesting works attempting to evolve ant pheromone communication, and others evolving swarm communication in general which is related closely enough.

A milestone early attempt to use a computer simulation to evolve ant foraging strategies using pheromones which resemble behaviours of biological ants was AntFarm (Collins & Jefferson, 1990a). AntFarm implemented an early form of neuroevolution, which was used to evolve the ANNs which learn behaviour for effective ant pheromone communication (Collins & Jefferson, 1990b). Neuroevolution methods in AntFarm evolved both the ANN connectivity pattern (topology) and weights of the ANN which were under genetic control in a genotype. Limitations were that: (1) AntFarm did not successful evolve any cooperative foraging which was the main objective. (2) A basic, conventional ANN was used, when compared to the wider range of operators, sigmoids and activation functions with complexification as used in more recent neuroevolution models such as NEAT (Stanley, 2004). (3) The number of neurons and connections were not under genetic control.

The first research to evolve Ant pheromone foraging was by Panait and Luke (2004).

More recently, Beem (2017) attempted to use NEAT to evolve the controller for individual agents in a swarm. However the methods failed to produce any ability for agents to find food, or communicate, or exhibit any swarm intelligence whatsoever. The most advanced behaviour that his agents ended by evolving was to walk in circles. Perhaps that was due to the coordinate system used, or a lack of random or sin wave inputs. The inputs to the NEAT ant controller included the ant's own position; the

¹ Dept. of Computer Science, Univ. of Chester, UK. Email: n.vaughan@chester.ac.uk.

intensity of pheromones at its location; whether or not the ant is carrying food at a given moment and the distance to the nearest food from two different points on the ant (for triangulation). The NEAT outputs are the agent's forward movement speed, its steering direction and the intensity of the pheromones it leaves behind. All agents within a swarm have the same neural network as controller.

Yong and Miikkulainen (2009) found that for cooperative tasks such as chase and evade, evolving several autonomous, cooperating neural networks to control a team of agents is more efficient and robust than evolving a single centralized controller. This potentially may apply in ants where two distinct roles are required – searching for food and returning to the nest.

Other attempts to evolve Swarm Intelligence using NEAT have failed, for example Chang & Worlanyo (2015) didn't see any communication being evolved. In other work, to some extent evolving swarm communication has succeeded (Floreano et al., 2007, Marocco and Nolfi., 2003, Yong and Miikkulainen., 2009). Rawal et al. (2012) successfully evolved cooperative communication between a group of predators who can only catch prey by communicating information codes to each other. A related work has evolved ants nest site localisation (Marshall, 2003).

Ant algorithms are generally most widely known through the wide literature on optimisation problems with ant colony (ACO) by Dorigo et al., (2006). Differing from this research, ACO algorithms conventionally must be hard coded by a designer and not evolved automatically.

Letendre and Moses (2013) used genetic algorithms to show that ant foriaging is improved in random food distributions and using both pheromone and site fidelity foriaging strategies. However their GAs were used only to adjust a set of parameters affecting behaviour, not to learn the behaviours themselves, which were hard-coded and pre-existing.

3 STATE OF OUTSTANDING PROBLEMS

There are some state of art current outstanding problems specifically within the evolving pheromone communication, some of which are addressed in this work.

Collins & Jefferson (1990a) suggested future work should involve: (1) a systematic study of the effect of food distribution on the evolution of foraging strategies, testing the model of Johnson et al. (1987). (2) evolution of foraging strategies that are strongly affected by competition to see if competitor colonies sharing a single environment will interfere with each others strategies, disrupting communication by overwriting pheromone or misleading trails – which is related to Anti-pheromone which was later separately used by Panait and Luke (2004). (3) Investigate previous suggestions that pheromone evolution requires incremental changes to vary the environment, slowly making foraging more difficult over time.

Future work can also focus on the limitations of Panait and Luke (2004) which was suggested as future work. (1) When using multiple food sources which decay when eaten, this results in a dynamic changing environment and this makes pheromone evaporation more important. (2) How does pheromone navigation change with introduction of predators. (3) Future work can investigate ants which can produce more than 2 pheromones, so the ants can also learn complex tours with multiple way-points and self-intersecting paths.

4 SYSTEM COMPONENTS

There are five components in the system which occur when an ant makes a move.

1. Pre-computed Inputs (ant sensors).

Ants have 13 input sensors: (1, 2) the location within the 9 adjacent cells (Moore neighbourhood) of the highest pheromone, (3, 4) the location within Moore neighbourhood which is closest to the nest, given by a 'compass sensor'. (5, 6) location within Moore neighbourhood of food. (7, 8) the direction of the ant's previous move, (9, 10) a direction picked at random, (11) a Boolean indicating whether the ant is currently carrying food, (12) a random number, (13) a fixed value of 1 (Bias). These are referred to as the pre-computed inputs and they remain the same even when the controller is changed (BM1, ANNs, NEAT).

Having a compass avoids the requirement to use two different pheromones. Compass is calculated by Pythagoras theorem using the x and y differences between ant and nest. In a grid system following the compass does not produce a direct path, it results in diagonal motion followed by perpendicular motion.

All of the 5 direction pre-computed inputs are represented in a consistent manner using two variables for x and y. These represent the change required in the ant's current x and y coordinates. These variables can be positive, negative or neutral. If both are neutral the ant would stand still (which would never be beneficial when foraging). If both were negative, the ant would move diagonally towards the origin (NW). With this method, the two variables can encode any direction within the 9 squares of the ant's Moore neighbourhood. If the ant chooses to follow the compass, it would then ignore the pheromone and vice versa.

2. Controller.

The controller is a 'black box' brain which decides the animal behaviour at timestep t, based on the pre-computed inputs from the ant's sensors. The experiments were repeated using different controllers: a hard-coded benchmark (BM1), a fixed topology neural network and neuroevolution by adjusting the topology and weights of an ANN.

3. Outputs.

The resulting output of the controller determines the direction in which the ant moves.

4. Post move local updates.

After each ant has moved, a number of post-move local updates are automatically applied. (1) If the ant is now standing on food and isn't carrying any, it automatically picks food up. (2) If the ant is carrying food, pheromone is deposited with strength inversely proportional to the time since collection. (3) If an ant is already carrying food and is now standing on a nest, it automatically drops the food. This representation realistically assumes that biological ants already could pick up and drop food before they evolved pheromone communication. These tasks are regarded as automatic responses which we assume have been learnt previously.

5. Global updates

After a full iteration, when all ants have finished making a move, a global update is triggered in which all pheromone is evaporated (decremented). A number of different evaporation rates including decrementing and various percentage reductions were tested to identify how evaporation rate affects the ability to evolve navigation controllers.

5 THE MACHINE LEARNING TASK

The given inputs and expected outputs were kept strictly equal for all tested controllers. Therefore here we can formally define the machine learning task based on the relationship between inputs and outputs of the controller. This is critical step because small changes to the representations of input and output can make big changes to the difficulty of the task for the controller to learn.

Inputs:

In total the task has 13 inputs: 10 (5 pairs of) input direction variables, 1 boolean input, 1 random number input and 1 fixed value (Bias). There are 2 outputs: x and y.

Of the 13 inputs 10 inputs are positional change inputs. These are in 5 pairs of x and y, relative to ants current position, to reach the optimal square within Moore neighbourhood for the 5 precomputed inputs: food, pheromone, compass, same-move or random-move. These all have three possible values -1,0,1.

```
Closestnestx
Closestnesty
Foodherex
Foodherey
Highestpheromonex
Highestpheromoney
rand_x
rand_y
same_x
same_y
```

For food and pheromone, 0,0 only occurs when none is found, which means that there is no need for having separate Booleans indicating food and pheromone presence. For compass, 0,0 only occurs when standing on the nest (in which case compass would not be useful as the ant would not be carrying food because it would have been dropped automatically).

There is one random number input called r. This is independent of random direction inputs. This is required so that ants can randomly determine when to move randomly.

r

A Boolean is included to represent whether food is currently being carried. This is an important flag because it defines one of two current states: (1) searching for food, or (2) bringing food back to nest. This information is not available in other inputs.

carryingfood

There are only two outputs. They represent the relative step the ant will take on this timestep. They can be a value from the set $\{-1,0,1\}$. Therefore the output of the controller purely determines the position of the ant's next move, which has 3^2 options, one for each square in the ant's Moore neighbourhood.

output_x
output_y

6BENCHMARK ALGORITHM BM1

The developed system included designing a custom developed hard-coded benchmark algorithm (BM1) for pheromone based food foriaging behaviour, shown as pseudocode in Fig. 1. The BM1 algorithm was used in this research as a comparison or gold standard to assess the performance of the fixed topology and evolving ANN algorithms.

The benchmark BM1 does produce efficient foriaging behaviour and also demonstrates that the pre-computed inputs provide all required information to complete the foriaging task. The benchmark performance was measured and used to evaluate the performance of ANN driven behaviour controllers which later evolved. The pseudocode gives a description of what happens for each ant to decide which direction to move in at each timestep. This implements two modes: searching for food, and retrieving food based on the carryingfood Boolean flag.

```
if (carryingfood){
   //follow compass to nest
}else if (food in neighbourhood){
   //step onto the food
}else if (pheromone in neighbourhood){
   //step onto strongest pheromone
}else if (rand%100<90){
   //continue previous direction
}else{
   //use a completely random direction.
}</pre>
```

Fig. 1. Pseudocode for the Benchmark Algorithm (BM1).

In the BM1 benchmark, when a random direction has been chosen, on consecutive timesteps, the direction has 90% chance of remaining constant. This means that ants travel largely in straight lines, broken by abrupt changes on 10% of steps. This causes the ants to more effectively spread out and cover the whole grid more quickly. The main benefit is that ants then have a much higher chance of running into an existing pheromone trail. This outperforms total random movement, whereby ants often retrace their steps in consecutive turns which results in a lack of general directed movements. Also the random direction is chosen from an 8-square Moore neighbourhood – so that standing still is never chosen as it would have no benefit.

It can be seen that efficient pheromone communication (BM1) can be captured in this simple pseudocode which consists of only 5 IF statements, plus the defined actions to perform within each condition. The machine learning task is to replicate the behaviour of these 5 IF statements and associate the correct actions with each case, by using only the 12 given inputs. This summarises the difficulty of the learning task. If the machine learning fails, it must be because the IF statement structure was too complex to learn, or the actions were not associated with the

correct conditions. The BM1 already demonstrates that the given 12 inputs are satisfactorily informative to complete the foraging task.

In the event of machine learning failure, that could be investigated in terms of machine learning complexity, rather than anything specifically about the foriaging task, because it could be assumed that other machine learning tasks with the same level of complexity would equally fail to be learnt.

7 RESULTS OF BM1 IN FIXED LEVELS

For a comparison between controllers (BM1, ANN, Neuroevolution), one fixed level was used. The obstacles and food were located in the same places. That ensured that each controller was subject to the exact same challenge. The fixed level is shown in Fig. 2.



Fig. 2. This shows the layout of the fixed level used for evaluating and comparing a variety of controllers.

When the ant colony was controlled by the BM1 algorithm and foriaged within this fixed layout level (Fig. 2.), the ant colony makes very consistent progress every time it is run (Fig. 3.). The small variations are due to the random movement of ants, taking slightly different times to first discover food sources before they are subsequently attended by large recruited swarms.

In total the fixed level happens to have 6630 foods. On a typical run in this fixed level, as those shown on the graph Fig. 3, BM1 collected the first food after 106 timesteps. At 2500 timesteps, 3143 foods were collected. By the time the run was halted at the 5000th timestep, food was still being actively collected, in total 4852 foods had been collected, so 1778 foods remained uncollected.

The behaviour of the BM1 can be further analysed by looking at Fig 4. which shows how often each direction was chosen. Standing still is the rarest move and diagonal bottom-right to top-right the most frequent. Also Fig. 5. shows how often each of the 5 IF statements from the pseudocode (Fig. 1) were triggered. Continuing in the same direction is the most frequent action and stepping onto food is the rarest action.

8FIXED NEURAL NETWORK TRAINING

Using a typical run of BM1 in a fixed level, as in Fig. 1. a training set for an ANN was produced.







Fig. 4 How often each Moore neighbourhood direction was chosen.



Fig. 5. How many times BM1 triggered each IF statement from the pseudocode.

This was achieved by writing to file all of the inputs and the resulting outputs for 10 ants over a 5000 timestep run. This data set contained 50,000 instances each containing the full set of 13 inputs and 2 outputs.

The training set was then used as the training set for a neural network. The aim was to identify whether an ANN could use backpropagation to learn the relationships between the inputs and the output produced by BM1. A 90% split was used to split into a 45,000 instance training set and an unseen 5,000 instance test data.

In order to use a single ANN to produce two outputs: x and y together, outpux_x and output_y were combined into a single output class with 9 values A-I (Fig 6). In total the ANN had 12 inputs and 9 output nodes, one for each class. With no hidden layers, the network classified 87.5% correctly. With one hidden layer of 10 nodes the MLP correctly classified 90.4% of test data. With two hidden layers of 10 nodes each (Fig. 6), accuracy improved to 92.3%. Training time increased with hidden layers.

It is not known if this trained ANN would in fact perform well as a controller for the ant simulation, or not. The misclassified instances could include important classes. It is hard to identify which situations the network failed in and if those would be critical or not to foriaging behaviour.



Fig. 6. The fixed topology ANN classified the optimal ant direction 92.3% of the time.

In order to clarify further, a training set was created with all the same inputs, but with 5 possible output class values, representing the 5 IF statements used in the pseudocode (Fig. 1). There are 5 pairs of directional inputs and the purpose of the 5 IF statements is to choose which of those 5 directions to follow (see pseudocode in Fig. 1). This test can clearly identify whether the 5 IF statements were correctly learnt, without having to also learn the correct actions to take within each IF statement. With one hidden layer, a fixed ANN was created and trained (Fig. 7). The ANN reliably identified the correct one of five IF statements 99.98% correctly classified. Only 1 of the 5000 was incorrectly classified.



Fig. 7. Learning the 5 IF statement classes accuracy 99.98%.

To summarise this fixed ANN training section, in this approach BM1 output data was used to train an ANN by back propagation. This step was useful to demonstrate that an ANN is capable of performing this task when properly trained, so therefore the task should not be overly complex for an evolving neural network.

This result suggests that a trained fixed topology ANN can perfectly learn to recognise which of the 5 IF statements should be triggered given the inputs. A harder learning task is to also determine which actions to take when each of the 5 IF statements are triggered, the ANN achieved this with 92.3% accuracy.

It is recognised that this approach differs from biological ants with natural selection, which have no pre-existing data to train the ants. In evolution, skills must be evolved without training or guidance, not towards a particular aim or objective. Therefore, the next section focusses on unsupervised evolution.

9EVOLVING ANN CONTROLLERS

Neuroevolution was applied to evolve neural networks which were then applied as the controller for ants. All ants in a colony had the same controller at each generation. But between generations, the controller was subject to genetic change, by modifying the ANN both in terms of the weights and the topological structure, including the number and location of connections.

Initially, the ANNs were set blank, with no hidden layer nodes. The additional nodes are added by evolution over time.

The fitness function was set to 1 point for each food picked up, 50 points for each food returned to the nest.

A comprehensive set of tests was done with 25 ants per colony, left to run for 900 timesteps. The ANN controllers were subject to neuroevolution in populations of size 15 organisms over 100 generations and this was repeated 10 times. Afterwards a further 5 repetitions of 100 generations was completed, this time with populations of 150 organisms which is a more conventional population size for neuroevolutionary algorithms.

In all runs food collection was learnt almost perfectly. By the 10th generation organisms often had peaked at a fitness of 25, meaning that every single ant successfully collected a food. In most experiments, the ANNs started to learn to return the food to the nest, which begun producing fitness of 62 in generation 13. At generation 42 the fitness was 2049, so the majority of ants were returning multiple foods to the nest. This cannot be explained by random movement alone which does not result in any food being returned. In the third run, the highest fitness reached 5059, meaning that 109 foods were collected, of which 99 foods were returned to the nest in only 900 timesteps, a highly efficient result, that means every ant on average collected food and returned it to the nest 4 times, outperforming BM1.

Visualisation of the evolved ANN structure (Fig. 8) shows that it had an additional 7 nodes had evolved, and 18 new connections, plus all of the weights throughout had evolved to optimal values.

It is hard to visualise why this ANN works so well because ANNs are a black box solution, yet some evolved nodes seem to make sense. The node in the bottom right was added by neuroevolution. It has connections to the Boolean flag pherom_here (input 14) and the pheromone x flag (input 5).



Fig. 8. The evolved ANN which has fitness 5059 and outperformed the benchmark algorithm. Red nodes were added by the neuroevolutionary algorithm, the two yellow nodes are outputs, one for the x movement, one for y movement. The green squares are the 14 ANN inputs.

That could make sense that the Boolean flag pherom_here could trigger the hidden neuron to send pheromone x information to the x output 1 only when present. Another observation is that most evolved nodes that connect to the x output do have connections coming from x inputs and the same is true for evolved nodes connected to the y output.

Subsequently the same experiment was repeated four times with a larger population size of 150. The runs produced fitnesses of 8533, 6178, 6083 and 2152. When controlled by the ANN with highest fitness 8533 the ants had collected 183 foods and returned 168 of those to the nest. Given the size of food clusters on the fixed map are over 1000 each, the score could be achieved by discovering a single food cluster.

For comparison, the BM1 was run 100 times for 900 time steps with 25 ants and the colonies had returned to the nest a range of food from 2 to 107, with an average of 60 foods. A possible reason why neuroevolution outperformed BM1 could be that it was overfitted to the test level layout.

Comparisons using purely random movement with 25 ants over 900 timesteps showed that a maximum of only 1 food was collected by ants, and no food was ever returned to the nest.

10 ALTERNATIVE SCENARIOS

A scenario was tested in which there was no compass and 1 pheromone. The compass input was experimentally removed, to identify whether a benchmark could be programmed without a 'nest compass' a sensor to nest direction, using only one pheromone. The algorithm would not correctly operate, because after finding the food there was no way for the ants to find a way back to the nest, so ants would move randomly, leaving

pheromone all over the place attracting other ants in the wrong directions (fig. 9).



Fig. 9. The BM1 algorithm running without a compass sensor – ants have no way of finding the nest once food is discovered and pheromone is scattered randomly.

Food distance from nest has various effects. With closer food, the pheromone trail will be stronger and it takes less time to get back to the nest. But longer trails have greater chance of other ants walking into them by accident, so further food may attract more ants that way. This is shown in Fig. 10. Two foods were discovered: a small food in the upper right is favoured compared to a larger food in the bottom left, because it is closer, ther pheromone is stronger and all ants abandon the larger food until the pheromone evaporates and knowledge of it's location is lost to the swarm. In Fig. 11b, the same affect is shown. Ants recruited to the bottom two foods only collected food once and when they reached the nest, they chose to follow the top food because that pheromone was stronger and the top food quickly depleted.



Fig. 10. A large food supply (lower left) is abandoned and forgotten in favour of a small food (upper right), because it is closer to the nest, causing a stronger pheromone trail.

Each decision that an ant makes can be subject to random probability so that ants are always capable of doing something unpredictable at any time. The effect of introducing a probability of random decisions is shown between Figs 11a and 11b.



Fig. 11. (a) With 0% chance of random behaviour at each timestep. White: ants. Red: food source. Blue-black: pheromone strength. Yellow: obstacle. Central point: nest. (b) With 70% chance of random behaviour at each timestep.

11 COMMON PROBLEMS ENCOUNTERED

One problem with using a compass to return to the nest is that if obstacles block the route in a 'v' shape, the ant never get past. In Fig. 12 this has occurred and long after the bottom two food sources are completed, the blocked food source hasn't been exploited yet. If two pheromones were used this may be avoided.



Fig. 12. Using one pheromone with compass, ants often get stuck behind a 'v' shaped yellow obstacle.

12 PHEROMONE STRENGTH DECREMENT

Hill climbing strategy has two main requirements. When an ant discovers a pheromone trail, there are usually two directions it could be followed. The ant should choose which direction using a hill climbing strategy to more easily find the food. That assumes that the strongest end of the trail will lead to the food. However, the whole trail evaporates over time, so the strongest end of each pheromone trail will naturally tend to be the end nearest the nest because the trail near the food has had a longer time to evaporate. Therefore, if hill climbing is to work, (1) the pheromone strength deposited by ants must decrease on consecutive squares as they get further from the food. Also (2) the evaporation rate must be slower than the reduction in strength left by ants on consecutive squares. This can be instructed either automatically, or it could be part of the ant's behaviour controller which is required to evolve. That would increase the search space for the ANN and would require two additional inputs (1) the number of steps taken since food and (2) the strength of pheromone already on this square from other ants and an output for the pheromone strength to deposit.

Problems can occur when ants are laying a pheromone trail and they walk across a separate pre-existing pheromone trail of a different strength (Fig. 13). In this case, ants should not be able to add to existing pheromone, up to a maximum limit on each square. They should also not be able to cause existing pheromone to reduce by overwriting pheromone left by other ants with a lower value. This causes problems when two paths from two food sources combine into one, and the ants from either food source will have taken a different number of steps and laying different strengths of pheromone. (Fig. 13.) The correct behaviour is that they should reset the square to their own calculated limit, unless a higher value is already present in which case they leave it as is.



Fig. 13. When two paths combine, pheromone should not be overwritten by ants who have travelled on a longer path.

13 CONCLUSIONS & FUTURE WORK

This paper has investigated the benefits of neuroevolution (NEAT) compared to fixed topology ANN by testing how pheromone behaviour can evolve in both, in relation to a hard coded designed benchmark (BM1).

Future work could investigate simulating other colony or swarm intelligences with communication. Examples include smells in the air or environment such as territory marking, or sounds used for predator detection, warnings or communication.

This paper has demonstrated neuroevolution applied to evolve pheromonome communication in simulated ant colonies. The core intelligence required to perform pheromone communication was summarised in form of the hard coded benchmark BM1, comprised of an IF block with 5 conditions. The 5 conditions were learnt by a fixed ANN and the actions to take within each IF statement were learnt with 92% accuracy.

This paper was organised into several stages: (1) Developed a benchmark algorithm which produces swarm food foriaging behaviour. (2) Used the benchmark to produce a training dataset linking the ant sensor inputs to the desired output movement direction. (3) The training set was used to train a fixed topology neural network which produced the desired output in 92% of cases. (4) Implemented neuroevolution to evolve an ANN with augmented topology and weights to produce foriaging behaviour. This was successful and the evolved ANN controller resulted in high number of foods being collected by the swarm and returned to the nest. The evolved controller outperformed the benchmark algorithm which presumably was due to overfitting to a fixed level layout.

REFERENCES

m

- Gaggioli. Optimal Experience in Ambient Intelligence. In: Ambient Intelligence. G. Riva, F. Vatalaro, F. Davide, M. Alcañiz (Eds.). IOS Press (2005).
- [2] Jin, A., & Austin, M. (2014). Comparative Evolutions of Swarm Communication.
- [3] Marocco D, Nolfi S. "The Emergence of Communication in Evolutionary Robots". In: Philosophical Transactions of the Royal Society London A.361 (2003), pp. 2397–2421.
- [4] Baddeley, B., Graham, P., Husbands, P., & Philippides, A. (2012). A model of ant route navigation driven by scene familiarity. PLoS computational biology, 8(1), e1002336.
- University Of Sussex, 2006, How ants find their way, Science Daily, https://www.sciencedaily.com/releases/2006/10/061018094651.ht
- [6] Collins, R. J., & Jefferson, D. (1990a). Antfarm: Towards simulated evolution. Computer Science Department, University of California.
- [7] Marshall, J., Kovacs, T., Dornhaus, A., & Franks, N. (2003). Simulating the evolution of ant behaviour in evaluating nest sites. Advances in Artificial Life, 643-650.
- [8] Hamilton, W. D. (1964). The genetical evolution of social behaviour. Ii. Journal of theoretical biology, 7(1), 17-52.
- [9] Wolfram S, A new kind of science.
- [10] Adler, F. R., & Gordon, d. M. (2003). Optimization, conflict, and nonoverlapping foraging ranges in ants. The American Naturalist, 162(5), 529–543.
- [11] Doursat, R., Sayama, H., & Michel, O. (2013). A review of morphogenetic engineering. Natural Computing, 12(4), 517-535.
- [12] Wohlgemuth, S., Ronacher, B., & Wehner, R. (2001). Ant odometry in the third dimension. Nature, 411(6839), 795–798.
- [13] Hölldobler, B. (1976). Recruitment behavior, home range orientation and territoriality in harvester ants, Pogonomyrmex. Behavioral Ecology and Sociobiology, 1(1), 3–44.
- [14] Thiélin-Bescond, M., & Beugnon, G. (2005). Vision-independent odometry in the ant Cataglyphis cursor. Naturwissenschaften, 92(4), 193–197.
- [15] Beem FV, 2017, Analysis of NEAT and application in swarm intelligence, Amsterdam University, Undergraduate Dissertation, Supervisor: Rein van den Boomgaard.

- [16] Conforth and Meng, Toward Evolving Neural Networks using Bio-Inspired Algorithms
- [17] Chang R, Worlanyo S, 2015, Evolving Swarm Communication with NEAT.
- [18] Rawal, A., Rajagopalan, P., Miikkulainen, R., & Holekamp, K. (2012). Evolution of a communication code in cooperative tasks. Artificial Life, 13, 243-250.
- [19] Collins, R. J., & Jefferson, D. R. (1990b). An artificial neural network representation for artificial organisms. In International Conference on Parallel Problem Solving from Nature (pp. 259-263). Springer, Berlin, Heidelberg.
- [20] Panait, L., & Luke, S. (2004). Learning ant foraging behaviors. In proceedings of the Ninth International Conference on the Simulation and Synthesis of Living Systems (ALIFE9) (pp. 575-580).
- [21] K. Letendre and M. E. Moses. Synergy in ant foraging strategies: Memory and communication alone and in combination. In Proceedings of the 15th Annual Conference Companion on Genetic and Evolutionary Computation (GECCO '13 Companion), pages 41–48, New York, NY, 2013. ACM.
- [22] T. Paz Flanagan, K. Letendre, W. Burnside, G. M. Fricke, and M. Moses. How ants turn information into food. In 2011 IEEE Symposium on Artificial Life (ALIFE 2011), pages 178–185, Piscataway, NJ, 2011. IEEE Press.
- [23] Hecker, J. P. (2015). Evolving Efficient Foraging Behavior in Biologically-Inspired Robot Swarms (Doctoral dissertation, The University of New Mexico).
- [24] Stanley, K. O. (2004). Efficient evolution of neural networks through complexification (Doctoral dissertation).
- [25] Yong, C. H., & Miikkulainen, R. (2009). Coevolution of role-based cooperation in multiagent systems. IEEE Transactions on Autonomous Mental Development, 1(3), 170-186.
- [26] Floreano, D., Mitri, S., Magnenat, S., & Keller, L. (2007). Evolutionary conditions for the emergence of communication in robots. Current biology, 17(6), 514-519.
- [27] Dorigo, M., Birattari, M., & Stutzle, T. (2006). Ant colony optimization. IEEE computational intelligence magazine, 1(4), 28-39.
- [28] Johnson, L. K., Hubbell, S. P., & Feener Jr, D. H. (1987). Defense of food supply by eusocial colonies. American Zoologist, 27(2), 347-358.

Kinetic Models of Relevant Phenomena Related to Opinion Dynamics

Stefania Monica and Federico Bergenti¹

Abstract. This paper outlines recent analytic results regarding the dynamics of the opinion in large multi-agent systems. Discussed results are based on a unifying framework which takes inspiration from mathematical kinetic theories to describe analytically the long-time asymptotic properties of the opinion in large multi-agent systems. Different models to describe the effects of interactions among agents are considered to take into account different sociological phenomena that are known to govern the dynamics of the opinion. For each of such models, the proposed framework allows the analytic derivation of the dynamics of collective properties, such as the average opinion and its variance. Discussed results are in accordance with known empirical evidence, and their analytic form contributes to a solid theory of opinion dynamics.

1 Introduction

Even if there is no agreement on a formal definition of what a multiagent system is, in this work multi-agent systems are considered as *large* groups of agents which interact in *intelligent* ways. Note that multi-agent systems are assumed to be made of a large number of agents, so that the behaviours of single agents is less interesting than the behaviour of the multi-agent system as a whole. In addition, note that intelligence is associated with interactions rather than with agents, so that the focus is put on the *collective intelligence* [31] of the multi-agent system as a whole rather than on the possible intelligence of single agents.

Since in this work multi-agent systems are assumed to be composed of a large number of agents, the detailed description of the properties of single agents is unfeasible. For this reason, the results described in the rest of this paper are focused on the study of collective properties of a multi-agent system, which can be broadly defined as properties that characterise the multi-agent system as a whole. Traditionally, the study of such collective properties is performed using simulation, and many results concerning the dynamics of multiagent systems are obtained by directly simulating interactions among agents (see, e.g., [8]). The major drawback that is traditionally associated with such simulation approaches is that the validity of results obtained by simulations depends on the choices of the specific parameters used to perform simulations and, hence, it is limited to simulated scenarios. Even if simulated scenarios can be considered representative of other scenarios, there is often no clear identification of the generality of results obtained by simulations. For this reason, it is of interest to study analytic models to describe the dynamics of multi-agent systems. The results of analytic models are often more general than the results of simulations. In addition, their generality is clearly identified by the assumptions taken in the development of the model. Note that analytic models are typically simpler than those considered in simulations, but they guarantee more robust results. Finally, note that analytic results can be verified by simulation in interesting cases.

The analytic study of the collective properties of multi-agent systems can be performed using a framework inspired from mathematical kinetic theories (see, e.g., [4]). Such a framework can be used to study the asymptotic characteristics of collective properties of multiagent systems under the assumption that studied systems are completely decentralized and with no form of supervised coordination. According to the proposed framework, observable collective properties of a multi-agent system can be derived analytically from the description of the effects of single interactions among agents. In this work, interactions are limited to message exchange which involve only two agents. Time is modelled as a sequence of discrete steps, which may not have the same duration, and each step corresponds to a single interaction among two agents. No restriction is imposed on the topological properties of the multi-agent system, and each agent is free to interact with any other agent. Note that mathematical kinetic theories are commonly described as generalisations of classic kinetic theory of gases. Such an approach is certainly possible, but it is worth noting that the class of systems, either physical or not, that can be described using the generic framework of mathematical kinetic theories is much wider that gases. Mathematical kinetic theories can be used to study all systems in which the major cause of evolution is given by interactions, and the system is so large that a statistical approach is justified.

In the rest of this work, it is assumed that each agent is associated with a scalar value intended to quantify the opinion of the agent regarding a fact. The adopted framework inspired from mathematical kinetic theories is used to study the long-time asymptotic dynamics of collective properties of the opinion in multi-agent systems. Therefore, discussed results can be broadly considered as part of *sociophysics* [9], which is an emerging discipline that uses models and tools from physics to study sociological phenomena. In addition, it is worth remarking that the ideas behind the adopted framework are not limited to the study of opinion dynamics and the proposed approach can be applied to describe other collective properties (see, e.g., [10, 30] for examples relative to economy).

This paper is organised as follows. Section 2 briefly recalls relevant sociological phenomena. Section 3 outlines the major elements of the adopted framework based on mathematical kinetic theories, and it summarises major results obtained from the application of the framework. Finally, Section 4 concludes the paper and outlines future research directions.

¹ Dipartimento di Scienze Matematiche, Fisiche e Informatiche, Università degli Studi di Parma, Parco Area delle Scienze 53/A, 43124 Parma, Italy {stefania.monica, federico.bergenti}@unipr.it

2 Relevant Sociological Phenomena

The study of the dynamics of the opinion is a traditional topic of the research on multi-agent systems. It has been faced using a number of different approaches and it originated a large body of literature. In addition, in recent years, its importance increased rapidly because of the explosive and pervasive diffusion of online social networks.

Models designed to describe the effects of social interactions among agents are not only studied in the context of multi-agent systems, but they are also central to various fields of sociology (see, e.g., [13]). In particular, starting from empirical investigations, sociologists intend to derive scientific theories and models to explain collective behaviors and socio-cultural differentiation [15]. The central idea of most of proposed model is the sociological phenomenon called *compromise* [1], which is the tendency of individuals to change their opinions towards those of others they interact with, trying to reach consensus. According to the idea of compromise, consensus among all individuals in a society should be reached after a sufficiently large number of interactions.

It is well known that empirical observations [3] do not confirm the long time asymptotic property of compromise, so to find models to explain pluralism, sociologists have described a variety of phenomena related to opinion formation. One of such phenomena is homophily [29], according to which individuals tend to communicate with those with similar opinions. In order to properly integrate homophily in models, bounded confidence models are considered. Such models assume that two individuals interact only if their opinions do not differ more than a given threshold. Bounded confidence models allow explaining opinion clustering, which is used to justify the existence of groups of individuals with similar opinions inside the same group, and also to account for the existence of different opinions among different groups [12]. Another important sociological phenomena, intended to avoid monoculture is negative influence, according to which individuals evaluate other agents on the basis of some parameters, and they only interact with those with positive scores. Negative influence can explain bipolarism [14]. According to sociologists, however, the idea of negative influence is not supported by empirical evidence [15] and, for this reason, other phenomena have been considered, such as opinion noise. Opinion noise is modeled as a random additive quantity which leads to arbitrary opinion changes with small probability [11]. Moreover, in [16], the idea of striving for uniqueness is introduced to avoid the convergence of opinions to a single value. This is related to the idea that single individuals may want to get distinguished from others and, therefore, they may change their opinions if too many individuals share the same opinion. This idea is opposed to the idea of fashion trends [2]. Finally, the sociological phenomena according to which the opinion of each agent can be influenced by the social context is known as diffusion [8].

We have already proposed analytic models to study, under proper assumptions, all mentioned phenomena [7, 17–21, 24, 26, 28] and to derive properties concerning the temporal evolution of the opinion in the considered system, in terms of the average opinion and its variance. Among the wide variety of studied models, this paper focuses on compromise, which is a key ingredient of all sociological models intended to describe the dynamics of the opinion in a society. Note that the adoption of the general framework of mathematical kinetic theories ensures that studied models can be combined into complex models which consider various phenomena. Actually, all mentioned phenomena can be modelled as additive terms of the interaction rules adopted to describe how single interactions modify the opinions of interacting agents.

3 Kinetic Models of Opinion Dynamics

In order to apply the general framework of mathematical kinetic theories to study the dynamics of the opinion, a number of assumptions on studied multi-agent systems and on the sociological phenomena considered in the formation of the opinion are needed. A few assumptions have been already mentioned in the introduction of this paper, and they are rephrased here to make this section self-contained.

The general framework of mathematical kinetic theories can be used to study the asymptotic characteristics of collective properties of multi-agent systems under the assumption that studied multi-agent systems are composed of a fixed number of $n \in \mathbb{N}$ interacting agents. In addition, studied systems are assumed to be completely decentralized and with no form of supervised coordination. Observable collective properties of a multi-agent system can be derived analytically from the description of the effects of single interactions among agents. Interactions are limited to message exchange among two agents. Time is modelled as a sequence of discrete steps, which may not have the same duration, and each step corresponds to a single interaction among two agents. No restriction is imposed on the topological properties of the multi-agent system, and each agent is free to interact with any other agent.

Opinion has been modeled in the literature both as a discrete [32] and as a continuous [12] variable. While discrete models are typically used to address situations where a finite number of options are available, e.g., in political elections, continuous models are typically used to study opinions related to a single topics, varying from strongly disagree to completely agree. In the adopted framework, the opinion is modelled as a continuous variable $v \in \mathbb{R}$. With no loss of generality, throughout the paper it is assumed that $v \in I = [-1, 1]$, so that ± 1 represent extremal opinions and values close to 0 correspond to moderate opinions.

Finally, opinion models differ because of the sociological phenomena adopted to describe how the opinions of agents change after single interactions. Opinion may change because of a variety of phenomena, which are briefly recalled in previous section to set the accepted nomenclature of sociology. Single models typically include a specific subset of such phenomena depending on the specific characteristics of the studied multi-agent systems, on targeted application scenarios, and on the background of authors. The proposed framework can accommodate all mentioned phenomena, but in this work the focus is on compromise.

The results discussed in the rest of this section are organised according to the common approach of identifying two types of models. In the first type, called homogeneous models, all agents are considered equal and agents can be treated as indistinguishable. Such models are commonly used to study multi-agent systems that are so large that the individuality of single agents does not contribute to the collective properties of the system. Mathematically, such models are easier to study and they form the basis for the second type of models, known as heterogeneous models. In heterogeneous models, agents are grouped into disjoint classes on the basis of some characteristics. The characteristics used to group agents into classes are arbitrary, but it is common to group agents on the basis of the ways they interact with other agents. The classification of agents allows accommodating different types of agents into the models, and it allows accounting for agents with diverse propensity towards other agents. Note that the general framework of mathematical kinetic theories allows classes to collapse to singleton sets, which allows single agents to be modelled as an individuality. In such an extreme case, each agent can be associated with a specific behaviour, but the framework is still capable of providing results on collective properties of the multi-agent system as a whole. Note that the specific description of each and every single agent is practically feasible only if the multi-agent system is composed by a limited number of agents. Therefore, heterogeneous models with singleton classes are used when the size of the studied multi-agent system is sufficiently small to model each agent in isolation, and sufficiently large to justify the statistical approach of mathematical kinetic theories.

3.1 Homogeneous Models

In the case of homogeneous models, mathematical kinetic theories postulate the existence of a distribution function f(v, t) whose temporal evolution is described according to the *Boltzmann equation* (see, e.g., [23, 24]). The distribution function f(v, t) represents the number of agents with opinion in (v, v + dv) at time t, and it is defined for each $t \ge 0$ and for each $v \in I$. According to the definition of f(v, t), by integrating f(v, t) over I, one obtains the number of agents n, which is assumed to be known and fixed

$$n = \int_{I} f(v, t) \mathrm{d}v. \tag{1}$$

Similarly, the average opinion of the multi-agent system can be computed as

$$u(t) = \frac{1}{n} \int_{I} v f(v, t) \mathrm{d}v.$$
⁽²⁾

In addition, it is also possible to compute the variance of the opinion of the multi-agent system as

$$\sigma^{2}(t) = \frac{1}{n} \int_{I} (v - u(t))^{2} f(v, t) \mathrm{d}v.$$
(3)

In agreement with mathematical kinetic theories, we assume that the distribution function f(v,t) evolves according to an integrodifferential equation which is a generalization of the Boltzmann equation. More precisely, homogeneous models assume that the temporal evolution of f(v,t) can be described using the following equation

$$\frac{\partial f}{\partial t}(v,t) = \mathcal{Q}[f](v,t) \tag{4}$$

where the left-hand side represents the temporal evolution of the distribution function and the right-hand side is called *collisional operator* to adopt the nomenclature of mathematical kinetic theories. The explicit expression of the collisional operator depends on the specific rules which govern interactions among agents and it ultimately embeds studied sociological phenomena in the model.

Following a common approach of mathematical kinetic theories, rather than studying the Boltzmann equation (4), its *weak form* is considered. The weak form of a differential equation is obtained by multiplying both sides of the considered equation by a smooth function with compact support, denoted as *test function*, and then integrating the obtained equation. Hence, the weak form of the Boltzmann equation (4) is

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{I} f(v,t)\phi(v,t)\mathrm{d}v = \int_{I} \mathcal{Q}[f](v,t)\phi(v,t)\mathrm{d}v \tag{5}$$

where the right-hand side is called *weak form of the collisional operator* Q with respect to test function $\phi(v, t)$. Functions that satisfy (5) are called *weak solutions* of the Boltzmann equation. Proper choices of test functions can be used to describe the temporal evolution of collective properties of multi-agent systems. In detail, the use of (5) allows studying

- 1. The average opinion of the multi-agent system, using test function $\phi(v) = v$; and
- 2. The variance of the opinion of the multi-agent system, using test function $\phi(v,t) = (v u(t))^2$.

In order to exemplify the analytic results that can be obtained using the described framework (see also previous works [5,6,22,23,25, 27]), we briefly summarize the major results obtained in the study of a very simple model of opinion dynamics which, among the variety of sociological phenomena cited in previous section, considers only compromise. Note that compromise is the key ingredient of all models of opinion dynamics and it is based on the idea that the opinions of two agents becomes closer after each interaction. Mathematically, denoting as v and w the pre-interaction opinions of two agents and as v' and w' the post-interaction opinions of the same agents, compromise corresponds to the following inequality

$$|v' - w'| \le |v - w|.$$
(6)

In order to ensure that compromise is properly studied, the following interaction rules can be used

$$\begin{cases} v' = v - \gamma(v - w) \\ w' = w - \gamma(w - v) \end{cases}$$
(7)

where γ is a parameter of the model that quantifies the effects of compromise. The value of γ measures the propensity of an agent to change its opinion in favor of the opinion of another agent. Note that in order to properly model compromise, namely to satisfy (6), the values of γ must be chosen in the interval $I_{\gamma} = (0, 1)$. As a matter of fact, it is easy to prove that values in I_{γ} guarantee that (6) is satisfied. Using the interaction rules in (7), right-hand side of the weak form of the Boltzmann equation can be written explicitly as

$$\beta \int_{I^2} f(v,t) f(w,t) (\phi(v',t) - \phi(v,t)) \mathrm{d}v \mathrm{d}w \tag{8}$$

where β represents the probability of interaction among two agents.

Under proper assumptions detailed in [24], the study of the solutions of the weak form of the Boltzmann equation can be used to prove relevant properties of the long-time asymptotic behaviour of the average opinion. In detail, the use of $\phi(v) = v$ in the weak form of the Boltzmann equation can be used to show that the average opinion of the multi-agent system is conserved. In addition, the use of $\phi(v,t) = (v - u(t))$ allows proving that all agents tend to the same long-time asymptotic opinion (see [24] for further details).

As an illustrative example, Figure 1 shows the values of the variance of the opinion in a homogeneous system where interactions among agents are governed by the rules defined in (7). Such results are obtained by considering a system composed of 10^3 agents and by assuming that $\gamma = 0.1$. Figure 1 shows that the variance of the opinion exponentially tends to 0 as the number of interactions increases, thus leading to consensus.

3.2 Heterogeneous Models

When agents cannot be considered equal with respect to their attitude towards interactions, previous homogeneous models cannot be applied. The rest of this section is devoted to extend previous results to cases where agents can be classified into disjoint groups with respect to their specific behaviours at interactions. Resulting heterogeneous models can be used in more realistic situations, and they can be also adopted in the extreme case of singleton classes. In such a case, each



Figure 1. Values of the average opinions $\{u_s(t)\}_{s=1}^6$ as a function of the number of interactions for the parameters discussed in Section 3.2.

agent has specific characteristics that describe its behaviour during interactions with other agents. Note that even if such a case could be ideally used to model all multi-agent systems, it is feasible only when the number of agents is sufficiently small to ensure that every agent can be described independently.

The assumptions and the nomenclature of heterogeneous models do not differ from that of homogeneous models. The only difference is that agents are assumed to be statically classified into $m \in \mathbb{N}$ classes, with $m \leq n$. Such a difference imposes that heterogeneous models need m distribution functions $\{f_s(v,t)\}_{s=1}^m$, each of which represents the number of agents of a single class with opinion in (v, v + dv) at time t. In agreement with the relative equation of the general framework of mathematical kinetic theories, the number of agents of class s at time t can be written as

$$n_s = \int_I f_s(v, t) \mathrm{d}v, \tag{9}$$

and it does not depend on time because the adopted classification is assumed to be static. The total number of agents is obtained by summing the number of agents of each class, as follows

$$n = \sum_{s=1}^{m} n_s. \tag{10}$$

Moreover, following the same approach used in the general framework of mathematical kinetic theories, one can compute the average opinion of agents of class s as

$$u_s(t) = \frac{1}{n_s} \int_I v f_s(v, t) \mathrm{d}v.$$
(11)

The definition of the average opinions of all classes allows computing the global average opinion of the system at time t as

$$u(t) = \frac{1}{n} \sum_{s=1}^{m} n_s u_s(t).$$
 (12)

Observe that the average opinion of the system is obtained by a properly weighed sum of the average opinions of all the classes. Finally, the variance of the opinion of agents of class s can be computed as

$$\sigma_s^2(t) = \frac{1}{n_s} \int_I (v - u_s(t))^2 f_s(v, t) dv.$$
 (13)

The general framework of mathematical kinetic theories assumes that each distribution function $f_s(v, t)$ evolves according to a balance equation, which is the analogous of the Boltzmann equation (5)

$$\frac{\partial f_s}{\partial t}(v,t) = \sum_{r=1}^m \mathcal{Q}_{sr}[f_s, f_r](v,t).$$
(14)

From (14) it is evident that the collisional operator relative to class s corresponds to the sum of the contributions $Q_{sr}[f_s, f_r]$ of the interactions between the agents of class s with the agents of class r.

The study of the *weak form* of the Boltzmann equation is useful to derive proper differential equations involving the temporal evolution of interesting macroscopic parameters of the system, namely the average opinion of each class and the variance of the opinion of each class. Just like for the case of homogeneous models, the weak form of (14) is obtained by multiplying it by a test function $\phi(v, t)$ and by integrating with respect to v. Hence, the weak form of the Boltzmann equation can be written as

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{I} f_s(v,t)\phi(v,t)\mathrm{d}v = \sum_{r=1}^m \int_{I} \mathcal{Q}_{sr}[f_s,f_r](v,t)\phi(v,t)\mathrm{d}v.$$
 (15)

In order to exemplify the analytic results that can be obtained using the described framework (see also previous works [5, 17–24], we briefly summarize the major results obtained in the study of a very simple model of opinion dynamics which, among the variety of sociological phenomena cited in Section 2, considers only compromise. Let us denote as s and r the generic classes of two interacting agents, whose pre-interaction opinions are v and w, respectively. The postinteraction opinions v' and w' of the two agents are computed as

$$\begin{cases} v' = v - \gamma_{sr}(v - w) \\ w' = w - \gamma_{rs}(w - v) \end{cases}$$
(16)

where $\{\gamma_{sr}\}_{s,r=1}^{m}$ are the deterministic parameters of the model. The value of a parameter γ_{sr} measures the propensity of an agent of class s to change its opinion in favor of the opinion of an agent of class r. Note that, just like for homogeneous models, in order to reproduce compromise, the values of $\{\gamma_{sr}\}_{s,r}^{m}$ must be chosen in the interval $I_{\gamma} = (0, 1)$.

As a matter of fact, using (16) it is possible to observe that (6) is equivalent to

$$|1 - (\gamma_{sr} + \gamma_{rs})| < 1$$

for all pairs of s and r, which is always satisfied if all parameters $\{\gamma_{sr}\}_{s,r=1}^m$ are restricted to I_{γ} . Using the interaction rules in (16), the right-hand side of the weak form of the Boltzmann equation for each class $s \in \{1, \ldots, m\}$ can be written explicitly as

$$\beta \sum_{r=1}^{m} \int_{I^2} f_s(v,t) f_r(w,t) (\phi(v',t) - \phi(v,t)) \mathrm{d}v \mathrm{d}w$$
(17)

where β represents the probability of interaction among two agents, as in the homogeneous model.

Under proper assumptions detailed in [23], the study of the solutions of the weak form of the Boltzmann equation can be used to prove relevant properties of the long-time asymptotic behaviour of the average opinion. In detail, under the assumption that $\gamma_{sr} = \gamma_{rs}$



Figure 2. Values of the average opinions $\{u_s(t)\}_{s=1}^6$ as a function of the number of interactions for the parameters discussed in Section 3.2.

for all pairs of s and r, the use of $\phi(v) = v$ in the weak form of the Boltzmann equation can be used to show that the average opinion of the multi-agent system is conserved. In addition, the use of $\phi(v,t) = (v - u_s(t))$ allows proving that all agents tend to the same long-time asymptotic opinion, regardless of their specific class (see [23] for further details).

Figure 2 shows illustrative results concerning a system composed of 10^3 agents, divided into 6 classes. In detail, Figure 2 shows the values of the average opinions $\{u_s(t)\}_{s=1}^6$ of each class, as a function of time (i.e., as a function of the number of interactions). Such results are derived on the basis of the analytic framework which is shortly outlined above and which is detailed in [23]. To obtain the results shown in Figure 2, the following parameters are considered. We assume that the 10^3 agents are divided into the 6 classes with:

$$n_1 = 9 n_2 = 150 n_3 = 400 n_4 = 300 n_5 = 140 n_6 = 1.$$
(18)

Let us remark that, according to (18), some classes are large (e.g, class 3 and 4), while other classes are composed of a few units of agents (class 1) or of a single agent (class 6). Concerning the values of the parameters that characterise compromise, we assume that $\gamma_{sr} = \gamma_s$, which means that the propensity of an agent to change its opinion through an interaction only depends on its class and not on the class of the interacting agent. In detail, the following values are considered:

$$\begin{aligned} \gamma_1 &= 0.1 \quad \gamma_2 = 0.2 \quad \gamma_3 = 0.3 \\ \gamma_4 &= 0.35 \quad \gamma_5 = 0.25 \quad \gamma_6 = 0.15. \end{aligned} \tag{19}$$

Finally, the initial distribution functions of the opinion are supposed to be the following:

$$f_1(v,0) = \mathcal{U}(-1,-2/3) \quad f_2(v,0) = \mathcal{U}(-2/3,-1/3) f_3(v,0) = \mathcal{U}(-1/3,0) \quad f_4(v,0) = \mathcal{U}(0,1/3) f_5(v,0) = \mathcal{U}(1/3,2/3) \quad f_6(v,0) = \mathcal{U}(2/3,1)$$
(20)

Figure 2 clearly shows that the values of the average opinion of different classes converge to the same value after a (relatively) small number of interactions.

4 Conclusion

This paper outlined how the general framework of mathematical kinetic theories can be applied to study the dynamics of the opinion in multi-agent systems. After a brief introduction intended to motivate the need of analytic models to study the dynamics of the opinion in multi-agent systems, the paper surveys the major sociological phenomena that has been already studied using the proposed approach. One of the most relevant of such phenomena is surely compromise, which is introduced by sociologists to model the fact that interactions sometimes make agents change their opinions to try to reach consensus. Compromise is a well understood phenomenon, both theoretically and empirically, so its description in terms of the proposed approach is valuable. The last section of the paper first outlined how the general framework of mathematical kinetic theories can be used to study the opinion dynamics, then it detailed a model of compromise and presented relevant results. Presented results are discussed for two classes of models: homogeneous models, which assume that all agents are equal, and heterogeneous models, which allow agents to have specific behaviours. Presented results are in accordance with the general understanding of compromise, and they can be used to study analytically compromise.

Planned future developments of the described research intend to expand the expressive power of proposed models by allowing parameters to be expressed in terms of random variables. Such an option is important to devise models whose results can be compared with empirical evidence in real scenarios, where the the behaviours of agents can be modelled only in terms of stochastic processes. In addition, another future development of the described research regards the application of the adopted approach to the study of other collective properties of multi-agent systems. Actually, the general framework of mathematical kinetic theories can be tailored to the specific needs of multi-agent systems to study a number of interesting properties, not strictly related to opinion formation.

REFERENCES

- [1] R. P. Abelson, *Contributions to Mathematical psycology*, Frederiksen N., Gulliksen H. eds., New York, 1964.
- [2] E. Abrahamson, 'Management fashion: Lifecycles, triggers, and collective learning processes', *Administrative Science Quarterly*, 44, 708–740, (1999).
- [3] R. Axelrod, 'The dissemination of culture: A model with local convergence and global polarization', *Journal of conflict resolution*, 41(2), 203–226, (1997).
- [4] A. Bellouquid and M. Delitala, *Mathematical Modeling of Complex Biological Systems*, Modeling and Simulation in Science, Engineering and Technology, Birkhäuser, Basel, 2006.
- [5] F. Bergenti and S. Monica, 'Analytic study of opinion dynamics in multi-agent systems with two classes of agents', in *Proceedings of* 17th *Workshop Dagli Oggetti agli Agenti* (WOA 2016), volume 1664 of *CEUR Workshop Proceedings*, pp. 17–22, Catania, Italy, (June 2016). ISSN: 16130073.
- [6] F. Bergenti and S. Monica, 'An analytic model of the impact of skeptical agents on the dynamics of compromise', in *Proceedings of* 18th Workshop Dagli Oggetti agli Agenti (WOA 2017), volume 1867 of CEUR Workshop Proceedings, pp. 73–78, (2017). ISSN: 16130073.
- [7] F. Bergenti and S. Monica, 'A kinetic framework to study opinion dynamics in multi-agent systems', in *Proceedings of the AISB Annual Convention*, The Society for the Study of Artificial Intelligence and the Simulation of Behaviour (AISB), pp. 65–70, (2017).
- [8] E. Bonabeau, 'Agent-based modeling: Methods and techniques for simulating human systems.', *Proc. Natl. Acad. Sci*, 7280–7287, (2002).
- [9] B. K. Chakraborti, A. Chakrabarti, and A. Chatterjee, *Econophysics and sociophysics: Trends and perspectives*, Wiley, Berlin, 2006.
- [10] S. Cordier, L. Pareschi, and G. Toscani, 'On a kinetic model for simple market economy', *Journal of Statistical Physics*, **120**, 253–277, (2005).

- [11] S. Galam, Y. Gefen, and Y. Shapir, 'Sociophysics: A new approach of sociological collective behavior', *Journal of Mathematical Sociology*, (2009).
- [12] R. Hegselmann and U. Krause, 'Opinion dynamics and bounded confidence models, analysis, and simulation', *Journal of Artificial Societies* and Social Simulations, 5(3), (2002).
- [13] P. J. Di Maggio and W. Powell, 'The iron cage revisited institutional isomorphism and collective rationality in organizational fields', *American Sociological Review*, **48**, 147–160, (1983).
- [14] N. P. Mark, 'Culture and competition: Homophily and distancing explainations for cultural niches', *Amm. Sociol. Rev.*, 68, 319–345, (2003).
- [15] M. Mäs and A. Flache, 'Differentiation without distancing. Explaining bi-polarization of opinions without negative influence', *PLOS One*, 8(11), (2013).
- [16] M. Mäs, A. Flache, and D. Helbing, 'Individualisation as driving force of clustering phenomena in humans', *PLOS One*, 6(10), (2010).
- [17] S. Monica and F. Bergenti, 'A stochastic model of self-stabilizing cellular automata for consensus formation', in *Proceedings of* 15th *Workshop "Dagli Oggetti agli Agenti"* (WOA 2014), Catania, Italy, (September 2014).
- [18] S. Monica and F. Bergenti, 'A kinetic study of opinion dynamics in multi-agent systems', in *Atti del Convegno* (AI*IA), Ferrara, Italy, (2015).
- [19] S. Monica and F. Bergenti, 'Kinetic description of opinion evolution in multi-agent systems: Analytic model and simulations', in *Proceed*ings of the 18th International Conference on Principles and Practice of Multi-Agent Systems (PRIMA), pp. 483–491, Bertinoro, Italy, (2015).
- [20] S. Monica and F. Bergenti, 'Simulations of opinion formation in multiagent systems using kinetic theory', in *Proceedings of* 16th Workshop "Dagli Oggetti agli Agenti" (WOA '15), Napoli, Italy, (June 2015).
- [21] S. Monica and F. Bergenti, 'An analytic study of opinion dynamics in multi-agent systems with additive random noise', in *Procs. Italian Conference on Artificial Intelligence* (AIXIA), volume 10037 of *Lecture Notes in Artificial Intelligence*, pp. 105–117, Genova, Italy, (2016).
- [22] S. Monica and F. Bergenti, 'A study of consensus formation using kinetic theory', in *Proceedings of the* 13th International Conference on

Distributed Computing and Artificial Intelligence (DCAI 2016), volume 474 of *Advances in Intelligent Systems and Computing*, pp. 213–221, (June 2016). ISSN: 21945357, ISBN: 978-331940161-4.

- [23] S. Monica and F. Bergenti, 'An analytic study of opinion dynamics in multi-agent systems', *Computers & Mathematics with Applications*, 73(10), 2272–2284, (2017).
- [24] S. Monica and F. Bergenti, 'Opinion dynamics in multi-agent systems: Selected analytic models and verifying simulations', *Computational & Mathematical Organization Theory*, 23(3), 423–450, (2017).
- [25] S. Monica and F. Bergenti, 'Outline of a generalization of kinetic theory to study opinion dynamics', in *Proceedings of the* 14th International Symposium on Distributed Computing and Artificial Intelligence (DCAI), volume 620 of Advances in Intelligent Systems and Computing, pp. 301–308, (2017).
- [26] S. Monica and F. Bergenti, 'An overview of kinetic approaches to study consensus formation', in *Proceedings of the AISB Annual Convention*, The Society for the Study of Artificial Intelligence and the Simulation of Behaviour (AISB), pp. 71–72, (2017).
- [27] S. Monica and F. Bergenti, 'Two analytic models of compromise in multi-agent systems', in *Proceedings of the* 15th German Conference on Multiagent System Technologies (MATES), volume 10413 of Lecture Notes in Artificial Intelligence, pp. 38–53, (2017).
- [28] S. Monica and F. Bergenti, 'A stochastic model of diffusion in opinion dynamics', in *Proceedings of the 10th International Conference on Agents and Artificial Intelligence*, volume 1, pp. 115–124, (2018).
- [29] A. Nowak, J. Szamrej, and B. Latan, 'From private attitude to public opinion: A dynamic theory of social impact', *Psycol. Rev.*, 97, 362– 376, (1990).
- [30] F. Slanina, 'Inelastically scattering particles and wealth distribution in an open economy', *Physical Review E*, 69, 46–102, (2004).
- [31] D. H. Wolpert and K. Tumer, 'Collective intelligence, data routing and Braess' paradox', *Journal of Artificial Intelligence Research*, 16, 359– 387, (2002).
- [32] E. Yildiz, A. Ozdaglar, D. Acemoglu, A. Saberi, and A. Scaglione, 'Noisy continuous opinion dynamics', *Journal of Statistical Mechanics*, 9, 1–13, (1982).

HILDA - A Health Interaction Log Data Analysis Workflow to Aid Understanding of Usage Patterns and Behaviours

Maurice D Mulvenna¹, Raymond R Bond¹, Alexander Grigorash¹, Siobhan O'Neill², Assumpta Ryan³

Abstract. Health and wellbeing products and services for individuals are becoming increasingly popular as people realise the benefits provided by lifelogging or quantified-self platforms in such areas as exercise, diet management and mood. However, in addition to the data that users record using these platforms, all user interactions and events can be elusively logged to represent usage. Such user interaction or event logs provide rich and large datasets that can fuel applied artificial intelligence. As products and services based on these digital interaction technologies are taken up across public healthcare provision, should healthcare policy and practice take more cognisance of the opportunities and risks in gathering interaction data? Is 'healthcare' ignorant that there is knowledge in such data? Are there differences between event logging in healthcare and other areas such as commerce, media and industry? In order to realise benefits in analysing such data, methods that help ensure consistency, accuracy, data protection, as well as reproducibility of knowledge derived from log data need to be examined. This paper presents methods to explore usage log data and a process workflow followed by a presentation of two real world case studies. The workflow has been coined Health Interaction Log Data Analysis (HILDA) and focuses on data prospecting and machine learning stages to show the opportunities realisable in analysing interactional or event data automatically recorded by digital healthcare services.

1 INTRODUCTION

Products and services based on digital interaction technologies typically include mobile device apps as well as browser-based apps to a lesser extent, and can include telephony-based services, text-based chatbots and voice activated chatbots. Many of these digital products and services are simultaneously available across many channels in order to maximise availability for users. The focus of this paper is to examine digital interaction technologybased products and services developed for use in the health and wellbeing domain, explore how these technologies can automatically log user interactions and events for subsequent analysis and business intelligence, and propose a methodology for data mining of digital products and service event logs in a healthcare context. Digital interaction technologies offer useful methods for real time data capture of the interactions of users with the products and services. We have become accustomed to using tools such as Google Analytics in order to generate aggregated usage reporting for our websites, but this kind of log analysis is available for all the digital interaction technologies we design. Indeed, we have the ability to design what data are recorded, how and where it may be stored, and crucially, how it can be analysed to reveal individual or collective usage patterns. This paper reports on two case studies in digital interaction technologies service analysis, where the usage data are examined using a structured data mining pipeline that is proposed and detailed in the methods.

The main focus of previous research on the analysis of usage logs for digital interaction technologies used in health and wellbeing has been to aid in usability analysis [1] or to reveal usage patterns in using technology [2]. Research has also been carried out to explore how rehabilitation devices can have data or event logging incorporated, but this has been more to support the goal of device monitoring [3]. More recent research has examined engagement data in web-based intervention platforms but has primarily focused on visualisation of the log data [4].

Our goal in incorporating usage or event logging or metadata logging is to provide relevant data sources as a basis for identification and exploration of individual and collective behavioural patterns of products and services developed using digital interaction technologies.

This work is potentially important since many national health departments including the National Health Institute (NHS) are looking to use digital technologies such as health apps for selfmanagement of diseases and thus logging user interactions would allow for greater insight into user needs and may provide ideas for improving these digital interventions, for example through enhanced personalisation. The NHS would benefit since the data can be automatically and hence cost-effectively collected. Such data may facilitate new ways for epidemiological analyses and provide data to inform health policies. If the NHS promote health apps and log analysis is insightful, then perhaps there is a need for a standard to maximise the utility of recorded event logs for analysis in healthcare contexts.

The structure of the paper is as follows. In the following section, methods encompassing usage log data types and process workflows is described. In the subsequent results section, the two case studies in the paper are introduced, before data prospecting and machine learning stages are described and results presented. The final sections of the paper provide for discussion and conclusions. We aim to examine the value and limitations of user log analysis in the healthcare domain.

¹ School of Computing, Ulster Univ., BT37 0QB, UK. Email: {md.mulvenna, rb.bond}@ulster.ac.uk

² School of Psychology, Ulster Univ., BT52 1SA, UK. Email: sm.oneill@ulster.ac.uk

³ School of Nursing, Ulster Univ., BT48 7JL, UK. Email:

aa.ryan@ulster.ac.uk

2 METHODS

A user event log must at least be comprised of three columns or variables as part of a tabular structure, comprising a unique identifier for the user (can be anonymous); the event that was recorded; and the date and time (preferable precision to seconds) of that event. This paper proposes a methodology for data mining such user event logs in a healthcare context (see Figure below). This study then takes a case study-based approach to examine the feasibility of using this model for analysing log data. Case study 1 involves an analysis of user event logs from a healthcare app that is used by people living with dementia and their carers for reminiscing about their past to encourage conversations, social connectedness, mutuality with their spouse and reflection as well as improving quality of life. This app was used in a 12 week trial during which all user events and features used were recorded. Log data includes the event/feature used (browsing photos, watching a video etc.), the date and time as well as a unique identifier for each user. Case study 2 involves analysis of ~3.5 million call logs made to a mental health and wellbeing helpline (Samaritans Ireland). Basic log data include time, date and duration of the call as well as a unique identifier for each caller. The results in this paper are very much autoethnographic of using our user log analysis workflow detailed in this paper. However, we do provide actual results of data analytic workflows in the case studies to provide an example of the insights revealed by the data mining process.

2.1 Proposed workflow model for log data analytics

Standardising workflows are crucial in order to ensure consistency and that best practices are adopted in a domain. A number of standards for carrying out a data science, data mining or a machine learning project have been proposed. For example, the Cross-industry standard process for data mining (CRISP-DM) is a data mining process model encompassing the following stages: Business understanding, Data understanding, Data preparation, Modeling, Evaluation, Deployment [5]. CRISP-DM has been available in various guides since 1996. An updated variant developed by IBM, called Analytics Solutions Unified Method for Data Mining/Predictive Analytics (ASUM-DM) [6] expands on CRISP-DM.

However, to date, there has not been a standardised workflow that embraces data mining in log analysis. Figure 1 below depicts Health Interaction Log Data Analysis (HILDA) workflow, which has been applied to two health and wellbeing log datasets: a mental health helpline call log dataset; and a user log dataset from a healthcare app to facilitate reminiscence for people living with dementia and their carers.

The first phase of the HILDA workflow model involves data cleaning. This comprises of data quality checks and imputations for missing data (NULL and NA fields). This is followed by a normalisation of the date stamps. That is to say that each date of each event is converted to a relative date for each user. For example, the first logged event for each user will be transformed to 'day 1', allowing like-for-like temporal analyses for all user logs. Moreover, seasonality is also often determined as a separate feature. The time stamps are converted to a simple integer from 0 to 23 denoting hour of the day. Oftentimes, classification of morning, afternoon and evening are used.



Figure 1. Workflow for Health Interaction Log Data Analysis (HILDA), involving three high level phases of data preparation, data prospecting and machine learning

Following data preparation, data prospecting is carried out. This comprises of analysing each variable for inherent characteristics by viewing histograms and probability density functions as well as identifying correlations via bivariate analysis. This is followed by time series analysis of a representative sample of users, or all if possible. This involves data visualisations of user events over their entire journey of using the service.

Following data prospecting is often a machine learning phase which can involve both unsupervised and supervised machine learning. Unsupervised machine learning comprises of extracting simple features that characterise basic usage behaviours such as frequency of use and number of interactions per week. These features are used in a clustering algorithm to determine the type of users the exist in using the service. This helps to understand the client base by behaviour. Having determined the type of service users that exist, one can proceed to predict service user type using features extracted from initial user behaviours, allowing for example the early detection of abstainers and adopters [7]. These predictive models can be developed using established supervised machine learning techniques such as decision trees and support vector machines which are very popular and arguably the most efficient methods that exist today.

The HILDA workflow model covers data preparation, data prospecting and machine learning in some detail. Deployment and usage is not explored in this version. Arguably, a focus on deployment was lacking in CRISP-DM as context for usage varies to such an extent that it is not helpful to be prescriptive about model deployment and usage.

3. RESULTS

The following sections outline the results and reflections of analysing user event log data in the health domain for each case study, describing the data prospecting and machine learning phases of the HILDA workflow in each.

3.1 Case study 1: Reminiscence health app user log data

This case study reports on the analysis of log data from a tablet application, specifically designed and developed to facilitate reminiscence for people with early to moderate stage dementia. Reminiscence is the sharing of memories relating to personal life experiences. It is the act of remembering and reflecting on real past events. The act of reminiscing can serve many functions that create bonds between people and in doing so, supports them to reflect on important life events and to attribute meaning to their lives [8]. The development of the app was a component part of a larger feasibility study to investigate the effects of individual specific reminiscence activity using a range of outcome measures, to explore users' views on the app; and to incorporate an economic analysis, examining the cost of implementing the app intervention in comparison with quality of life outcomes. The feasibility study incorporated a paired sample of 28 dyads (person living with dementia and their carer), and applied several scales at start, midand end-point of a 12-week use of the app in the homes of people living with dementia and their carers, with one-to-one interviews with participants carried out at the end of the 12 weeks.

Data prospecting

The app was designed to incorporate a logging facility for key events by users across 45 specific activities, covering five different types of events. The five different canonical events include: entry (Logging in), admin (Adding a photo, deleting an audio, etc.), reminiscing (Viewing a video, viewing a photo, etc.), in the moment (ITM) questions and exit (Logging out). Thus, the behaviour of users can be analysed within and across each usage session, over the 12-week trial. The ITM questions comprise items from the primary outcome measure for the study, the Mutuality Scale developed by Archbold et al. [9].

The data show that the app was primarily used for reminiscing as expected. A total of 71% of interactions from people living with dementia were within the reminiscing sections of the system whereas only 47% of interactions from carers were within the reminiscing sections (p < 0.001). It is reassuring that people living with dementia mainly used the system for reminiscing. Only carers could carry out 'Admin' events such as adding a photo, as mandated by their access rights set at login. It can perhaps be seen as a positive sign that carers generally added to the music, pictures and videos that were uploaded to the app prior to the intervention beginning, rather than simply browsing those already there. There were twice as many interactions with photographs in comparison to music and five times as many interactions with photographs in comparison to video by people with dementia using the app. Reminiscing, with its history in photograph-based memory books, has been more about the image than music, sound or video, and this effect may be what is being seen in this data [10]. What is also interesting in this data is the popularity of music to people living

with dementia. Again, this is known from the literature [11] and anecdotally from carers of people living with dementia but it is useful to see this behaviour replicated in this trial data. The most popular times that the dyads of people living with dementia and carers prefer to use the app peak around 11am, 3pm and 8pm. These times correspond to post-breakfast, post-lunch and postevening meal times. We also calculated the number of unique days in which users interacted with the system, and there is a significant statistical correlation between the number of days the carer interacted with the system and the number of days the dyad's corresponding person living with dementia interacted (r=5.77, p<0.001).

Machine learning

In this study, we used K-means clustering algorithm given it is the most widely used and established clustering algorithm in the unsupervised machine learning literature. Using the elbow method, we discerned that 4 is a reasonably small number of clusters that would provide reasonable resolution in terms of explained variability. Clustering was based on the following features five features: number of interactions by person living with dementia, number of daily interactions by person living with dementia, the mean usage interval by a user and the standard deviation of usage interval by a user.

Four clusters were revealed by the k-means algorithm. The first cluster, 'the hooked adopter' constituted one dyad, who fully adopted the system. They had 7.2 times more interactions than their carer. Whilst the person with dementia obsessively used the app, the carer showed a normal amount of usage, hence the person with dementia was independently dedicated. The hooked adopter and dyad uses the app for over half the days in a month (55% of days) and with little variability uses the app every two days. The second cluster, labelled the 'typical user' encompassing the plurality of users, where 12 dyads or 43% fall into this cluster, hence making them the most typical user. These people living with dementia user only have 1.7 times more interactions with the app than their carer. This indicates that these users have some dependence on the carer for app usage. This dyad uses the app 15% of days in a month. This dyad are unpredictable when they will use the app but on average interacts with it every 6.61 days (approximately once per week). The third cluster, labelled 'disengaged irregular user' encompassed 7 dyads or 25% of users. These users had 25% fewer interactions with the app than the carer. Whilst the people with dementia had fewer interactions than their carers, the carers had fewer interactions than other carers in all other clusters. These dyads use the app 9% of the days in a month. However, typically they can go for 20 days without using the app making them the least consistent users of the app. The final cluster labelled the 'well supported dependent user' encompassed 8 dyads or 29% of users, the second largest group of users. These users have 36% fewer interactions with the app than their carers. The carers are very enthusiastic and have more interactions than other carers in all other clusters but they seem to struggle to get people with dementia users to the same engagement level. Similar to the typical users in cluster 2, these dyads interact with app 16% of the days in a month and on average use the app every 6.97 days. This unsupervised learning provided clusters that were clear and transparent to the health science researcher involved in the project. The next stage in this work is

to seek to identify correlations between the post-trial interviews with the dyads and the clusters enumerated above.

3.2 Case study 2: Mental health helpline call log metadata

Helplines are key elements of mental wellbeing and suicide prevention efforts, however little is known about how these services are used. This study involves analysis of digital telephony data sourced from a mental health and wellbeing charity in Ireland. The charity operates a national helpline to provide emotional support to anyone in distress or at risk of suicide. Whilst support is also offered via SMS, email and face to face, 95% of the contacts remain via telephone. Data was provided for all calls made to the charity in the Republic of Ireland for almost a 4-year period (April 2013 to December 2016). A total of 3.449 million calls was analysed, amounting to 725 calls per 1,000 population.

Data prospecting

Novel data analytics and machine learning approaches were used to identify populations of callers based on caller behaviour variations. Key findings include the identification of five clusters of callers based on caller persistence, that persist regardless of which year or group of years is considered. The volume of calls exhibits strong intra-day and intra-week repetitive patterns, while intra-month repetitions are conspicuously absent. The influx of new helpline callers is remarkably stable, at a rate of about 1,200 per month. This rate remains constant over time, showing virtually no fluctuations across months or years. The observed probability distribution of call durations cannot be adequately explained using simple modelling techniques. The complexity of this distribution indicates that a mixture of distributions generated by several subpopulations of callers is being observed. The dataset comprises a number of fields, however only the following fields were used in this study: the date-time stamp of the call arrival precise to the last second; the Boolean engaged field meaning that the call was dropped with a busy tone; the answered flag meaning that the call was passed to a volunteer from the charity; the duration of the call in seconds; and the unique caller ID. The caller IDs allowed us to enumerate the callers uniquely while providing no personally identifying or sensitive details. The IDs were associated with most, but not all, call arrivals. About 20% of the calls had the caller ID missing. At a first glance, the duration of calls appears to follow an exponential decay distribution, where the volume of calls decreases at a rate proportional to call duration. However, the distribution follows a complicated decay pattern with a rate that associates with the call duration in a non-obvious manner.

Machine learning

In this study, cluster analysis involves grouping a set of objects (e.g., callers based on their attributes) in such a way that objects in the same group (called a cluster) are more similar to each other in comparison to other groups (clusters). Clustering was initially performed for 2013-2015 timespan, and then on the 2016 data to check for stability of our findings. Only the calls from callers with a unique identifier were used (44,613 callers collected in 2013–2015, increasing to 61,287 unique callers by the end of 2016).

Callers were clustered using 3 caller attributes: number of calls; mean call duration; and standard deviation of call duration. We selected these parameters due to their explanatory power: the number of calls a person makes indicates their frequency of help seeking behaviour; the mean call duration indicates call length; and the standard deviation of call durations indicates a person's variability and consistency in conversation length. We also used K-means clustering algorithm in this case study.

Five clusters were revealed by the k-means algorithm. The first cluster named 'Elite prolific callers' had the largest average number of calls per caller in the cluster and the smallest cluster size. A handful (fewer than 50 callers over the 4-year time span) of extremely prolific callers, responsible for 20% of the total call volume. They call thousands of times, each call on average lasts about 4 minutes, with a small minority of calls lasting 10 minutes. The second cluster named 'Typical callers' had the largest cluster size. The majority of callers who call 5-6 times and almost always have a short 3- to 4-minute conversation each time. This cluster accumulates 40 to 50 percent of all callers depending on the time slice under consideration. The third cluster named 'Standard prolific callers' had the second largest average number of calls per caller, middling average call duration and the largest unexplained variability encompassed by the cluster. About 12 to 15 percent of callers are prolific, each calling hundreds of times and having call durations that are moderate in length (from a few minutes to half an hour long). The fourth cluster named 'Unpredictable erratic callers' had the largest average standard deviation of the call duration. About 3 to 5 percent of callers whose call duration varies considerably, with some calls lasting 3 minutes and some up to 1 hour. The final fifth cluster named 'One-off chatty callers' had the smallest average number of calls per caller accompanied by the largest average call duration. About 13 percent of callers that only call 1-2 times, have a long 30 minutes to 1 hour conversation, and do not return for any sustained support (the operational opposite to prolific callers).

Data mining of the association of caller IDs with the volume and duration of calls revealed several caller clusters, each describing a distinctive behaviour type. The most striking of those clusters, Elite Prolific callers, encompasses a small number of caller IDs responsible for a substantial share of the total call volume that the charity receives. Early identification of callers of this type and routing their calls to specialized advisers provides insight in the modelling of healthcare service usage, offering insights for evidence-based practice and operational decisionmaking.

4. DISCUSSION

The work presented in this paper introduces a new data analytics process workflow, called Health Interaction Log Data Analysis (HILDA), designed to accommodate processing of user, system event and interaction data logged on products and services that use digital interaction technologies. The HILDA workflow is specifically designed for use in log data pertaining to health and wellbeing products and services. We are seeing a rapidly growing use by individuals of these health and wellbeing products and services as people realise benefits provided by lifelogging or quantified self platforms in such areas as exercise, diet management and mood. HILDA is designed to enable the capture of best practices in user, event and interaction log preparation, data prospecting and machine learning phases across a broad range of health and wellbeing usage scenarios. The workflow has been developed in response to the practical needs of our research to make use of more standardised approaches in data analytics research; but more broadly in recognition of the opportunities and threats involved as public health sector providers seek to make use of these kinds of technologies. Two case studies have been presented to illustrate the utility of the HILDA workflow in helping the transitioning from the data to actionable knowledge.

Workflows such as HILDA can help ensure reproducibility of findings as knowledge is derived from log data and also support consistency and accuracy. Broader topics such as data protection, compliance with ethical guidelines, etc., need to be accommodated within these kinds of workflows, especially as the General Data Protection Regulation EU (GDPR) (https://www.eugdpr.org/) comes into force in 2018. Other ethical concerns need to be discussed, for example the ethics of logging geolocation with each event or interaction with a smartphone health app. Moreover, can supposedly anonymous log analysis be used to identify individuals? Also, the ethics of consent or lack thereof for logging events need to be considered. Websites often log events and make use of logging tools like Google Analytics without obtaining consent from the user. However modern web browsers now seek consent for using cookies so the future for consenting on elusive data collection on the web has perhaps progressed.

The HILDA workflow describes three stages or phases, including data preparation, data prospecting and machine learning. Once the machine learning stage is complete, the results can lend themselves to being labelled and described in clear terms that are easily understood beyond the data science community. This is the case in our two case study examples. However, this is not always the case. Often the results of the machine learning stage if, for example generated using deep neural network methods offer little in terms of rationale for results, which can hamper take up, either because of the inherent complexity of the findings or the need for any deployment to be fair, accountable and transparent perhaps due to legal reasons. There are many other uses of log data not mentioned in this paper, for example, building a recommender engine to understanding the feature preferences for similar users, perhaps taking a collaborative filtering approach. Moreover perhaps a Markov model can be used for state predictions using a probability matrix for switching between states. In the second case study, we did use Fourier analysis to determine dominant frequencies of calls and to filter-out nondominant frequencies to elicit a more noise free model of call volume over time. Hence, there are many techniques that be applied to log analysis that are not necessarily represented in this paper.

The HILDA workflow could arguably benefit from a fourth phase or stage that helps signpost how the knowledge and or model generated can be taken up by the sponsor or community. Variously, this could be called deployment, uptake or operationalisation.

A major limitation of event log analysis is that it is very focused on quantitative analysis and requires significant interpretation. As such, there is no way of knowing why users take a certain journey or prefer a certain feature or even why they have certain patterns and frequency of use. Put differently, log analysis does not provide any qualitative feedback. However, this can be partly addressed using Ecological Momentary Assessment (EMA aka experience sampling) to add qualitative to the quantitative. EMA involves prompts to users often as pop-up questions during the service experience to gather user opinion or state. As a result, we would recommend considering EMA to augment user log analysis. However, EMA could be a distractor to the service itself so it needs to be carefully considered.

5. CONCLUSIONS

In summary, this paper presented a relatively unsophisticated model for carrying out user log analysis and applied this process in two healthcare services that automatically logged events. We found that that both case studies provided business intelligence and insights that are useful for understanding service users and for improving the service itself. We would support the integration of event logging in digital health given such data can be easily collected and analysed.

In addition to the work presented on the two case studies in this paper, the research team are exploring the analysis processes involved in carrying out data analytics in several other projects. In one of these, the data pertains to the use of an app by people with learning disabilities as they rate their transport or consumer experience in their daily life. This data therefore has additional locational information, and requires careful ethical considerations as the data relates to vulnerable users. Future development of the method therefore needs to address ethical aspects as a core part of the process, perhaps to be considered horizontally across all technical phases of the process. In another project, users access a maternal mental health app that helps them to compile and curate an event diary relating to mood in addition to completion of health and wellbeing scales. The data in the project can, when analysed, offer insights back to the community of users and so the process need to take account of deployment issues.

ACKNOWLEDGEMENTS

The Reminiscence 'Inspired' app was co-created and designed with help and input from the Alzheimer's Society, Reminiscence Network Northern Ireland and people living with dementia and their carers. The authors gratefully acknowledge their support and the funding support provided by HSC R&D Grant COM/5016/14 for the project. Financial support for the metadata study was provided in part by Samaritans Ireland together with Ireland's National Office for Suicide Prevention. Our thanks to the research teams who worked on the projects with us, including Ciaran Moore, Colette Ramsey, Cherie Armour for the Samaritans project, and Aideen Gibson, Claire McCauley, Liz Laird, Brendan Bunting and Finola Ferry for the Inspired project.

REFERENCES

- K. Miller, P.J. Woollam, G. Powell, D. Hitchings, D., and J. Stallard (2007) A Rehabilitation Device Data Logging System, Disability and Rehabilitation: Assistive Technology 2(1):9-14.
- [2] Vagner Figuerêdo de Santana, M. Cecília C. Baranauskas (2010) Summarizing observational client-side data to reveal web usage patterns. In Proceedings of the 2010 ACM Symposium on Applied Computing (SAC '10). ACM, New York, NY, USA, 1219-1223.
- [3] Daniel Woo, and Joji Mori. 2004. Accessibility: A Tool for Usability Evaluation. In book. Computer Human Interaction: 6th Asia Pacific Conference, APCHI 2004, Rotorua, New Zealand, June 29-July 2,

2004. Proceedings, edited by Masood Masoodian, Steve Jones, and Bill Rogers, 531-39. Berlin, Heidelberg: Springer Berlin Heidelberg.

- [4] Morrison C, Doherty G, Analyzing Engagement in a Web-Based Intervention Platform Through Visualizing Log-Data J Med Internet Res 2014;16(11):e252, DOI: 10.2196/jmir.3575.
- [5] Shearer C., The CRISP-DM model: the new blueprint for data mining, J Data Warehousing (2000); 5:13—22.
- [6] Jason Haffar (2015) Have you seen ASUM-DM?, SPSS Predictive Analytics, IBM, Located at: https://developer.ibm.com/predictiveanalytics/2015/10/16/have-youseen-asum-dm/, Last accessed 24 January 2018.
- [7] Thickett, J., 2006 Connecting Older People: Consumer Engagement with Digital Services, London, Ofcom.
- [8] Butler, R.N. (1963). The life review: an interpretation of reminiscence in the aged. Psychiatry, 26: 65-76.
- [9] P.G. Archbold B.J. Stewart M.R. Greenlick T. Harvath (1990) Mutuality and preparedness as predictors of caregiver role strain. Research in Nursing & Health. 13: 375-384.
- [10] T. Wright (2009) Drawn from Memory: Reminiscing, Narrative and the Visual Image, Proceedings of the First International Workshop on Reminiscence Systems (RSW-2009), Cambridge, UK, 5 September, 2009 pp:37-42.
- [11] A. Sixsmith, G. Gibson (2007) Music and the wellbeing of people with dementia, Ageing and Society, vol. 27, pp. 127-145.